INTERNATIONAL Trade

Trade integration and production sharing

A characterization of Latin American and Caribbean countries' participation in regional and global value chains

Dayna Zaclicever





INTERNATIONAL TRADE

Trade integration and production sharing

A characterization of Latin American and Caribbean countries' participation in regional and global value chains

Dayna Zaclicever





This document has been prepared by Dayna Zaclicever, Economic Affairs Officer of the Division of International Trade and Integration of the Economic Commission for Latin America and the Caribbean (ECLAC).

The views expressed in this document, which has been reproduced without formal editing, are those of the author and do not necessarily reflect the views of the Organization.

United Nations publication ISSN: 1680-872X (electronic version) ISSN: 1680-869X (print version) LC/TS.2017/161 Copyright © United Nations, December 2017. All rights reserved Printed at United Nations, Santiago, Chile S.17-01271

Applications for authorization to reproduce this work in whole or in part should be sent to the Economic Commission for Latin America and the Caribbean (ECLAC), Publications and Web Services Division, publicaciones@cepal.org. Member States and their governmental institutions may reproduce this work without prior authorization, but are requested to mention the source and to inform ECLAC of such reproduction.

Contents

Abst	ract .		7	
Intro	oduct	ion	9	
I.	Ver	rtical specialization-related intermediate exports	11	
II. Latin American and Caribbean countries' participation in international				
	pro	duction networks	13	
	Ā.	Trends and patterns of countries' vertical specialization	13	
	B.	Latin America and the Caribbean's forward participation in GVCs	17	
		1. Current linkages	17	
		2. Trade agreements and GVC participation	24	
		3. Potential further integration into GVCs		
III. Intra-regional production networks in Latin America				
	A.	Current linkages		
		1. Argentina		
		2. Brazil		
		3. Chile		
		4. Colombia		
		5. Costa Rica		
		6. Mexico	40	
		7. Peru		
	В.	Intra-regional export potential		
IV.	Co	ncluding remarks	57	
Refe	rence	es		
Anne	ex Ad	litional tables and figures	65	
Inter	natio	onal Trade Series: issues published	82	

Tables

Table 1	Latin America and the Caribbean: preferential trade agreements with members of GVCs
Table 2	Latin America and the Caribbean: RCAs in intermediate exports to GVCs, 2010-2011
Table 3	Latin America and the Caribbean: RCAs in intra-regional intermediate exports 2010-2011 45
Table A.1	Industry breakdown of OECD's ICIO tables
Table A.2	List of countries covered by the OECD's ICIO tables
Table A.3	Selected Latin American countries: intra and extra-regional preferential
	trade agreements
Table A.4	Latin America and the Caribbean: share in imported intermediate goods embodied in Latin American countries' gross manufacturing exports by source industry, 2011
Figures	
Figure 1	Imported intermediates embodied in gross manufacturing exports, 1995, 2008, 2009 and 2011
Figure 2	Selected country groupings: main imported intermediates embodied in gross manufacturing exports 1995 and 2011
Figure 3	Selected country groupings: imported intermediate goods embodied in gross
1 1801 0	manufacturing exports by trade agreement status with origin countries,
Figure 4	Selected importing regions: origin-country composition of vertical specialization-related
I Iguie I	intermediate importing frequency of the country composition of vertical spectralization related 18
Figure 5	Selected country groupings: Herfindahl-Hirschman indices of destination
8	market and sector concentration of vertical specialization-related
	intermediate exports, 1995-1996 and 2010-2011
Figure 6	Factory North America: geographical origin of main imported inputs embodied
0	in gross manufacturing exports, 1995 and 2011
Figure 7	Factory Europe: geographical origin of main imported inputs embodied in gross manufacturing exports, 1995 and 2011
Figure 8	Factory Asia: geographical origin of main imported inputs embodied in gross manufacturing exports, 1995 and 2011
Figure 9	Argentina: imported intermediates embodied in gross manufacturing exports by exporting industry, 1995 and 2011
Figure 10	Argentina: composition of imported intermediate goods embodied in gross
	manufacturing exports, 1995 and 2011
Figure 11	Brazil: imported intermediates embodied in gross manufacturing exports
Eigung 12	by exporting industry, 1995 and 2011
Figure 12	Brazil: composition of imported intermediate goods embodied in gross
Figure 13	Chile: imported intermediates embodied in gross manufacturing exports
Figure 15	by exporting industry 1995 and 2011
Figure 14	Chile: composition of imported intermediate goods embodied
I Iguie I I	in gross manufacturing exports 1995 and 2011 36
Figure 15	Colombia: imported intermediates embodied in gross manufacturing
1.801.0.10	exports by exporting industry 1995 and 2011 37
Figure 16	Colombia: composition of imported intermediate goods embodied
0	in gross manufacturing exports, 1995 and 2011
Figure 17	Costa Rica: imported intermediates embodied in gross manufacturing
÷	exports by exporting industry, 1995 and 2011
Figure 18	Costa Rica: composition of imported intermediate goods embodied
-	in gross manufacturing exports, 1995 and 2011
Figure 19	Mexico: imported intermediates embodied in gross manufacturing
	exports by exporting industry, 1995 and 2011

Figure 20	Mexico: composition of imported intermediate goods embodied	
-	in gross manufacturing exports, 1995 and 2011	41
Figure 21	Peru: imported intermediates embodied in gross manufacturing	
	exports by exporting industry, 1995 and 2011	42
Figure 22	Peru: composition of imported intermediate goods embodied	
	in gross manufacturing exports, 1995 and 2011	43
Figure 23	Latin America and the Caribbean (selected subregions and countries): vertical	
	specialization-related intermediate exports by technology content, 2011	59
Figure 24	Latin America and the Caribbean (selected subregions and countries): Herfindahl-	
	Hirschman index of sector concentration of vertical specialization-related	
	intermediate exports, 2010-2011	60
Figure A.1	Latin America and the Caribbean: Herfindahl-Hirschman index of destination	
	market concentration of vertical specialization-related intermediate exports,	
	1995-1996 and 2010-2011	67
Figure A.2	Latin America and the Caribbean: Herfindahl-Hirschman index of sector concentration	
	of vertical specialization-related intermediate exports, 1995-1996 and 2010-2011	68
Figure A.3	Argentina: geographical origin of main imported non-primary inputs embodied	
	in gross manufacturing exports, 1995 and 2011	71
Figure A.4	Brazil: geographical origin of main imported non-primary inputs embodied	
	in gross manufacturing exports, 1995 and 2011	72
Figure A.5	Chile: geographical origin of main imported non-primary inputs embodied in gross	
	manufacturing exports, 1995 and 2011	73
Figure A.6	Colombia: geographical origin of main imported non-primary inputs embodied	
	in gross manufacturing exports, 1995 and 2011	74
Figure A.7	Costa Rica: geographical origin of main imported non-primary inputs embodied	
	in gross manufacturing exports, 1995 and 2011	75
Figure A.8	Mexico: geographical origin of main imported non-primary inputs embodied in gross	
	manufacturing exports, 1995 and 2011	76
Figure A.9	Peru: geographical origin of main imported non-primary inputs embodied in gross	
	manufacturing exports, 1995 and 2011	77

Diagram

Diagram 1	Selected country groupings: imported intermediate goods embodied in gross				
	manufacturing exports by origin region, 1995 and 2011	.16			

Abstract

In the last three decades, the rise of global value chains (GVCs) created new growth opportunities for developing countries, by allowing them to participate in international trade without the need to develop vertically integrated industries at home. Latin American and Caribbean countries have largely remained on the periphery of this process, as exporters of primary products and natural resource-based manufactures. Yet, studies on these countries' participation in cross-border production sharing have been hampered by a lack of comprehensive data.

This document combines information from input-output tables with trade data to provide new evidence on Latin American and Caribbean countries' gross forward participation in international production networks (i.e., as exporters of intermediate products that are incorporated into other countries' exports). The data show a limited insertion into extra-regional value chains (with the main exception of Mexico), as well as significantly lower levels of intra-regional production integration than countries in other regions. However, both extra and intra-regional linkages vary greatly across countries.

Although the limited forward engagement of most Latin American and Caribbean countries in international value chains could be related to structural factors such as distance to the major GVC regions, and across the continent, the data presented in this document show the potential for further integration. However, in order to maximize the positive spillovers into the domestic economy (in terms of productivity gains, creation of high-quality jobs, skills and technology transfer to local firms, and greater integration of small and medium-sized enterprises), Latin American and Caribbean countries need to upgrade in international value chains. This would require coordinated plans across a number of policy areas, including trade, foreign direct investment, innovation and human capital formation.

Introduction

In the last three decades, the rise of global value chains (GVCs) created new trade and growth opportunities for developing countries. The rapid progress in information and communication technologies (ICT), along with the reduction in trade costs, enabled different tasks within a production process to be performed by geographically dispersed processing units, leading to the sharing of production between developed and developing economies. This so-called globalization's second unbundling (or North-South production sharing) allowed developing countries to specialize in specific segments of the value chain, and participate in international trade without the need to develop vertically integrated industries at home.¹

The available evidence suggest the existence of a regional bias in the process of international fragmentation of production, as it has largely taken place within groups of neighbouring countries and regional trade blocs (Johnson and Noguera, 2012; Estevadeordal et al., 2013; Baldwin and Lopez-Gonzalez, 2015). Moreover, GVCs are organized around three major manufacturing hubs in North America, Europe and East Asia, centred in the United States, Germany, and Japan and China, respectively. This regionalization of international production networks is mainly explained by the role of geographical distance in determining international transport costs and delivery times, as well as by the existence of regional trade agreements.²

Although limited, the existing evidence on developing economies shows great diversity in their degree of participation in GVCs. Many East Asian countries have successfully integrated into international production networks, mainly within the Asian region, by specializing in niches of the manufacturing process where they have a comparative or competitive advantage. In contrast, Latin American and Caribbean (as well as African) economies have largely remained on the periphery of GVCs, as exporters of primary products and natural resource-based manufactures (a traditional specialization pattern accentuated in the 2000s by the boom in commodity prices).

In spite of being interconnected by a number of trade agreements (including the Andean Community, the Southern Common Market (MERCOSUR), the Central American Common Market (CACM), the Caribbean Community (CARICOM), the more recent Pacific Alliance Agreement, and several bilateral preferential agreements), Latin American and Caribbean countries exhibit significantly lower levels of intra-regional trade and production integration than countries in other regions (OECD, 2015; Cadestin et al., 2016; ECLAC, 2016). Also, the trade agreements signed with extra-regional

¹ While in the first unbundling (up until the mid-1980s) international competition occurred mainly at the level of sectors (or firms), in the second unbundling it occurs at the level of production stages (or tasks) (Baldwin, 2006; 2013).

² Several studies have found evidence of a larger effect of geographical distance and preferential trade agreements on trade associated with international production networks than on overall trade or trade in final products (e.g., Miroudot et al., 2009; Gamberoni et al., 2010; Lopez-Gonzalez, 2012; Orefice and Rocha, 2014).

economies (including those linking Chile, Colombia, Peru and Central American and Caribbean countries with the United States or the European Union) do not seem to have contributed much to promote Latin American and Caribbean economies' participation in GVCs. The exceptions are Mexico and, to a lesser extent, some Central American countries, which have been relatively successful in joining (mostly labour-intensive assembly segments of) international production networks, in particular those centred in the United States.

Yet, studies on Latin American and Caribbean countries' participation in international value chains have been hampered by the limited availability of data. Although conventional (i.e., gross) trade statistics on intermediate goods (or, more limited, parts and components) are used in the empirical literature as a proxy for GVC trade, they do not provide a link between the industry of origin and the industry actually using the intermediates in its production process. The use of input-output tables allows a more accurate identification of international production linkages. Particularly, inter-country input-output (ICIO) tables provide information on the domestic and foreign inputs used in the production by each industry, revealing the linkages between countries within international value chains. Unfortunately, the available ICIO tables cover only a few Latin American countries.³

A more comprehensive analysis of Latin America and the Caribbean's participation in international value chains can be approximated by combining information from input-output tables with trade data. This is the approach adopted in this document to examine countries' forward insertion into international production networks (i.e., as exporters of intermediate products that are incorporated into other countries' exports). Input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables are used to compute the value of imported inputs embodied in countries' gross manufacturing exports, by source industry. These inputs are then disaggregated by origin country to build a bilateral vertical specialization-related intermediate trade dataset, which is used to characterize countries' participation in cross-border production sharing along the period 1995-2011.⁴

The document is organized in four sections. Section I describes the construction of the database. Section II presents the general trends and patterns of countries' vertical specialization, to then focus on the analysis of Latin American and Caribbean countries' integration into the three major GVCs (the so-called Factory North America, Factory Europe and Factory Asia).⁵ Section III examines intra-regional production linkages within Latin America. Finally, section IV summarizes the main findings and draws some policy implications.

³ The Organisation for Economic Co-operation and Development (OECD)'s ICIO tables cover seven Latin American countries (Argentina, Brazil, Chile, Colombia, Costa Rica, Mexico, and Peru), while the World Input-Output Database (WIOD) only includes two of these countries (Brazil and Mexico). Other databases provide more extensive country coverage; however, the inclusion of countries with poor quality/non-official data results in a loss of statistical rigor. As for the South American Input-Ouput Table, developed by the Institute of Applied Economic Research (IPEA) of Brazil and ECLAC, it is not a global ICIO table (as it only includes the ten South American countries) and it is currently available only for 2005.

⁴ This approach is conceptually similar to that of Lopez-Gonzalez and Holmes (2011), although the implementation differs.

⁵ Factory North America comprises Canada, Mexico and the United States; Factory Europe includes the 28 current member countries of the European Union; while Factory Asia is given here by the 14 South and East Asian countries covered in the OECD's ICIO tables (i.e., Brunei Darussalam, Cambodia, China, Hong Kong, India, Indonesia, Japan, Malaysia, Philippines, the Republic of Korea, Singapore, Taiwan, Thailand, and Viet Nam).

I. Vertical specialization-related intermediate exports

One way of examining countries' participation in international production networks is through the concept of vertical specialization, introduced by Hummels et al. (2001). Vertical specialization measures link a country's imported inputs with its exports. Thus, they reflect the process by which, as a result of the geographical fragmentation of production, different countries become part of a single supply chain (De Backer and Yamano, 2012). Particularly, countries' forward participation in international production networks can be measured, in gross terms, by their exports of intermediate inputs that are incorporated into other countries' exports.

The measure of vertical specialization considered here includes both the direct and indirect (i.e., embodied in domestic inputs) foreign intermediates used by each country in the production of its exported manufactured goods,⁶ given by:

$$\mathbf{VS}_{c,t} = \mathbf{A}_{c,t}^{M} \left(\mathbf{I} - \mathbf{A}_{c,t}^{D} \right)^{-1} \hat{\mathbf{E}}_{c,t} = \mathbf{A}_{c,t}^{M} \mathbf{L}_{c,t} \hat{\mathbf{E}}_{c,t}$$
(1)

where the subscripts *c* and *t* denote country and time, respectively; $\mathbf{A}_{c,t}^{M}$ is the K×K matrix of direct imported input coefficients; $\mathbf{A}_{c,t}^{D}$ is the K×K matrix of direct domestic input coefficients (or technical coefficients); **I** is a K×K identity matrix; $\mathbf{L}_{c,t} = (\mathbf{I} - \mathbf{A}_{c,t}^{D})^{-1}$ is the K×K Leontief inverse; $\mathbf{\hat{E}}_{c,t}$ is a K×K diagonal matrix of gross exports by exporting industry; and K is the number of industries.⁷

Each $VS_{c,t}(i,j)$ (i.e., each component of matrix $VS_{c,t}$) is the value of the imported inputs from industry *i* used in country *c* at time *t* in the production of industry *j*'s gross exports. Thus, the imported intermediates from industry *i* embodied in country *c*'s total gross manufacturing exports is calculated as:

$$VS_{c,t}(i) = \sum_{j} VS_{c,t}(i,j)$$
⁽²⁾

⁶ In addition to the direct use of foreign intermediates, the production of exports requires the use of inputs sourced from domestic suppliers, which in turn may require the use of imported intermediates, and so on. Ignoring these indirect import requirements leads to an underestimation of the foreign content of exports.

⁷ Each coefficient of matrix $\mathbf{A}_{c,t}^{D}$, $\mathbf{a}_{c,t}^{D}(i,j)$, gives the value of products from domestic industry *i* used by industry *j* as intermediate inputs to produce one monetary unit of output. Similarly, each coefficient of matrix $\mathbf{A}_{c,t}^{M}$, $\mathbf{a}_{c,t}^{M}(i,j)$, shows the imported inputs from foreign industry *i* required by domestic industry *j* to produce one monetary unit of output. For details on these matrices see Ahmad et al. (2017).

The data used to compute $VS_{c,t}$ come from the 2016 edition of the OECD's ICIO tables, which provide information for 63 countries covering the years 1995-2011, with a 34-industry breakdown.⁸

The final step in the construction of the database is the disaggregation of $VS_{c,t}(i)$ by origin country:

$$VS_{p,c,t}(i) = VS_{c,t}(i) * m_{p,c,t}(i)$$
(3)

where $m_{p,c,t}(i)$ is the share of origin country *p* in country *c*'s total intermediate imports from source industry *i* at time *t*. This share is computed using 6-digit Harmonized System (HS) level data from the *Base pour l'Analyse du Commerce International* (BACI) developed by the *Centre d'Études Prospectives et d'Informations Internationales* (CEPII).⁹ The correspondence between industries and HS codes considered here is based on the conversion tables used in the construction of the OECD's ICIO tables. This ensures a good match between product-level bilateral trade and the industry of origin in VS_{c.t}(i).

The resulting vertical specialization-related intermediate trade database comprises 210 exporting/origin countries (including 33 from the Latin American and Caribbean region) and 63 importing/destination countries (7 of which are from the Latin American region), providing information at the industry level. The main advantage of this dataset is its broad country coverage, relative to the input-output data from which VS_{c,t}(i) is obtained. This allows a more comprehensive identification of the trade relationships associated with countries' vertical specialization. Although ICIO tables provide information on the geographical origin of the inputs imported by each country, they only identify those imports originated from the countries included in the tables (and a residual Rest of the world). The disaggregation by origin country proposed here allows a more complete analysis by including, in a consistent manner, all input-exporting countries. However, in terms of importing countries, the analysis can only include those for which input-output data are available (i.e., the 63 countries covered in the OECD's ICIO tables).

It should be noted that the data approach adopted in this document does not address the doublecounting problem that affects international trade statistics, which do not distinguish between the domestic and foreign value added embodied in traded products.¹⁰ An analysis based on measures of trade in value added would allow quantifying the actual contribution of each country to the value generated in each production chain. However, the (statistically rigorous) data required for such analysis —related with each country's production structure and the countries from which it supplies— are not available for most Latin American countries. Thus, the vertical specialization-related gross flows computed here should be considered a proxy for inter-industrial interactions between countries within international value chains. Notwithstanding this, they improve the characterization of countries' engagement in cross-border production sharing, relative to the use of conventional intermediate trade data.¹¹

⁸ The OECD's ICIO tables are available for downloading at http://www.oecd.org/sti/ind/inter-country-input-output-tables.htm. See tables A.1 and A.2 in the annex for a list of the industries and countries covered, respectively.

⁹ BACI provides bilateral values and quantities of exports, reconciling exporting and importing countries' data from the United Nations Commodity Trade Statistics Database (COMTRADE) (see Gaulier and Zignago, 2010).

¹⁰ Intermediate inputs and the goods in process in which they are subsequently embedded cross several national borders as they move along the supply chain. The fact that conventional trade statistics record the gross cumulated value embodied in traded goods, not just the value added in the exporting country, results in a double-counting problem that overstates the domestic content of exports.

¹¹ Another drawback of the data approach adopted here is the proportionality assumption implied in the disaggregation of foreign inputs by origin country (i.e., the fact that countries' import shares do not differ by using industry). However, this assumption is standard in the construction of ICIO tables.

II. Latin American and Caribbean countries' participation in international production networks

A. Trends and patterns of countries' vertical specialization

The rise of GVCs can be illustrated by the increase in the imported intermediate inputs embodied in countries' gross exports over the period 1995-2011. As shown in figure 1.A, the share of foreign intermediates in the gross value of manufactures exported by the three major GVCs rose until 2008, particularly in the case of Factory Asia.¹² By 2011, vertical specialization levels were similar to those of 2008, after recovering from the decline caused in 2009 by the global economic crisis.¹³

At the aggregate level, the Latin American countries covered in the OECD's ICIO tables, excluding Mexico (i.e., Argentina, Brazil, Chile, Colombia, Costa Rica, and Peru; referred hereafter as Latin America (6 countries)), exhibit a considerably lower content of imported intermediates in their manufacturing exports than Factory North America, Factory Europe and Factory Asia. Notwithstanding this, their dependence on foreign inputs also increased significantly between 1995 and 2008.

Although regional aggregates are informative, they hide wide differences across countries (driven mainly by structural characteristics such as market size, level of development, and specialization patterns). Within Latin America (6 countries), Costa Rica has a considerably higher gross backward participation in international production networks, reflecting the larger share of non-natural resource-based manufactures in its exports (see figure 1.B). In contrast, the higher natural resource intensity of the other five countries' manufacturing exports results in lower levels of vertical specialization (as the production of these exports tends to use relatively few imported inputs).¹⁴

¹² Manufacturing exports refer hereafter to foreign sales from the manufacturing sectors listed in table A.1 in the annex, excluding Coke, refined petroleum products and nuclear fuel.

¹³ Some recent estimates indicate that international fragmentation of goods production would have stalled, or even reverted, since 2011 (Timmer et al., 2016; OECD, 2016).

¹⁴ As shown in section III, aggregate vertical specialization levels also hide large differences across exporting industries within countries.

Similarly, within each of the three major GVCs, countries exhibit significant differences in their gross backward participation levels. In Factory North America, the import content of the United States' gross manufacturing exports (21% in 2011) is considerably lower than those of Canada and Mexico (37% and 44%, respectively). For Factory Europe, in 2011 vertical specialization levels ranged between a minimum of 26% (Romania) and a maximum of 58% (Hungary), with an average of 38%. In Factory Asia the values averaged 40%, ranging between 18% (Japan) and 57% (Cambodia).



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables. ^a Latin America (6 countries) includes Argentina, Brazil, Chile, Colombia, Costa Rica and Peru. Mexico is included in Factory North America.

The composition by source industry shows that the foreign intermediate goods embodied in the manufactures exported by Factory North America, Factory Europe and Factory Asia originate mainly from medium-high or high-technology industries (including Computer, electronic and optical equipment; Chemicals and chemical products; Motor vehicles; Machinery and equipment, not elsewhere classified (n.e.c.); and Electrical machinery and apparatus, n.e.c.) (see figure 2). Some medium-low-technology industries (particularly, Basic metals and, to a lesser extent, Fabricated metal products) and the primary mining sector are also among the main suppliers of imported intermediates for the three GVCs.¹⁵

Medium-high or high-technology industries are also an important source of foreign inputs for Latin American countries' exports (particularly, Chemicals and chemical products; Machinery and equipment, n.e.c.; Computer, electronic and optical equipment; and Motor vehicles), although to a lesser extent than for Factory North America, Factory Europe and Factory Asia. On the other hand, inputs from the mining sector have a significantly larger participation for Latin American exports.¹⁶

¹⁵ The technology-intensity classification of manufacturing industries considered in this document is based on that of the OECD (see *ISIC Rev. 3 Technology Intensity Definition* (OECD, online document: www.oecd.org/dataoecd/43/41/48350231.pdf)). The group of low-technology industries comprises Food products, beverages and tobacco (ISIC 15-16); Textiles, textile products, leather and footwear (ISIC 17-19); Wood and products of wood and cork (ISIC 20); Pulp, paper, paper products, printing and publishing (ISIC 21-22); and Manufacturing n.e.c., recycling (ISIC 36-37). Medium-low-technology industries include Coke, refined petroleum products and nuclear fuel (ISIC 23); Rubber and plastics products (ISIC 25); Other non-metallic mineral products (ISIC 26); Basic metals (ISIC 27); and Fabricated metal products (ISIC 28). Finally, the group of medium-high or high-technology industries comprises Chemicals and chemical products (ISIC 24); Machinery and equipment, n.e.c. (ISIC 29); Computer, electronic and optical equipment (ISIC 30, 32, 33); Electrical machinery and apparatus, n.e.c. (ISIC 31); Motor vehicles, trailers and semi-trailers (ISIC 34); and Other transport equipment (ISIC 35).

¹⁶ It should be noted that the increase in the share of Mining and quarrying as origin of foreign inputs, in the four country-groupings considered, might be largely explained by the sustained boom in prices experienced in the 2000s by crude oil, natural gas and mineral ores. Moreover, GVC indicators (such as the vertical specialization measure used in this study) are affected by relative price movements, since they are computed from current-price data.

As shown in figure 2, services play an important role in international fragmentation of production, representing a significant source of foreign intermediates for the manufactures exported by the three major GVCs, as well as for Latin American countries. Unfortunately, the limited availability of data on bilateral trade in services prevents from including them in the more detailed analysis that follows in this document. Also, hereafter crude and processed fuels are excluded for not being considered of interest for this study.



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables.

^a Latin America (6 countries) includes Argentina, Brazil, Chile, Colombia, Costa Rica and Peru. Mexico is included in Factory North America.

The composition of (non-fuel) foreign intermediate goods by geographical origin shows the regional character of Factory Europe and Factory Asia (see diagram 1). In 2011, more than 60% of total imported intermediate goods embodied in these GVCs' gross manufacturing exports had an intra-regional origin. A high proportion of foreign intermediates in Factory North America's manufacturing exports was also imported from the region itself (i.e., Canada, Mexico, and the United States), but extra-regional countries accounted for the majority of those inputs (particularly, Southeast Asian countries).

The limited intra-regional production integration of Latin American countries is revealed by the relatively low participation of the region as origin of the imported intermediate goods used in the production of these countries' manufacturing exports (slightly over 25% in 2011, with the South American subregion accounting for the bulk of this share). The data presented in diagram 1 also show that Latin America and the Caribbean's participation as a source of foreign intermediate goods for Factory Europe and Factory Asia is

very low (2.4% and 3.8%, respectively, in 2011), and concentrates in South American countries. For Factory North America, the region's share is somewhat higher (6.1% in 2011).



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

Created with NodeXL Basic from the Social Media Research Foundation (http://nodexl.codeplex.com)

^a Latin America (6 countries) includes Argentina, Brazil, Chile, Colombia, Costa Rica and Peru. Mexico is included in Factory North America. The size of the circles reflects the share of each region as origin of foreign intermediates in 2011 (i.e., the percentages next to the name of each origin region). Percentages in brackets correspond to the shares in 1995. The boundaries and names shown and the designations used in this map do not imply any official endorsement or acceptance

on the part of the United Nations.

The expansion of GVCs has been associated with the proliferation of trade agreements, particularly those that go beyond traditional market access issues (by including provisions related to foreign direct investment (FDI), intellectual property rights, competition policy, product regulations, and customs procedures).¹⁷ This is reflected in the large and increasing proportion of countries' backward linkages that originates in trade partners with which some kind of trade agreement exists (see figure 3). For Factory Europe and Factory North America, in 2011 75% and 52%, respectively, of the foreign intermediate inputs embodied in manufacturing exports was imported from countries with which a trade agreement was in place. In both cases, the main origin of foreign inputs was largely the regional bloc itself (i.e., the European Union and the North American Free Trade Agreement (NAFTA), respectively); however, the share of extra-regional agreements has increased.

For Factory Asia, most backward linkages are not related with any trade agreement, but the proportion of inputs imported from a country with which there is an agreement increased significantly between 1995 and 2011 (from 6% to 27%). Also in this case, intra-regional agreements have the largest

¹⁷ As shown by Orefice and Rocha (2014), the relationship between international production networks and deep integration can go in both directions. On the one hand, deep trade agreements may facilitate trade amongst potential members of a supply chain, stimulating the creation of production networks. On the other hand, countries already involved in international fragmentation of production may be willing to sign deeper trade agreements with their partners in order to secure their trading relationships as suppliers of intermediate products.

participation. For Latin America (6 countries), trade agreement-related imports also account for an increasing share of the total (36% in 2011, compared to 11% in 1995), but the participation of intra and extra-regional agreements is the same (after a larger increase of the latter).





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI); and information on trade agreements from the World Trade Organization's Regional Trade Agreements Information System (RTA-IS). ^a Latin America (6 countries) includes Argentina, Brazil, Chile, Colombia, Costa Rica and Peru. Mexico is included in Factory North America. For Factory North America, Factory Europe and Factory Asia intra-regional agreements correspond to those

North America. For Factory North America, Factory Europe and Factory Asia intra-regional agreements correspond to those signed with other members of the GVC; for Latin America (6 countries), they include the trade agreements signed with any Latin American or Caribbean country. Partial scope agreements are excluded.

B. Latin America and the Caribbean's forward participation in GVCs

1. Current linkages

In line with their position in world trade, Latin American and Caribbean countries' forward participation in international production networks is limited. Their share in the foreign intermediate goods used in the production of world manufacturing exports was only around 5% in 2011, showing a small increase since 1995 (from nearly 4%). In addition, this participation is highly concentrated in a few countries, with Brazil, Mexico and Chile accounting for more than 70% of the total (see figure 4). Moreover, the six main origin countries (Brazil, Mexico, Chile, Argentina, Peru, and Costa Rica) explained 90% of the region's gross forward linkages in 2011.¹⁸ For Factory North America, most Latin American intermediate goods come from Mexico, while Brazil and Chile are, within the region, the main sources for Factory Europe and Factory Asia.

¹⁸ Latin America and the Caribbean's participation in world goods exports, excluding fuels, was 6.4% in 2011. This participation was somewhat less concentrated than that of the region's forward linkages, as five countries (Mexico, Brazil, Argentina, Chile, and Colombia) accounted for around 70% of the total (with the first two countries explaining more than 50%).





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Percentages in brackets next to the name of each importing region indicate the share of Latin America and the Caribbean in the region's total vertical specialization-related intermediate imports in 2011.

In terms of destination markets, the concentration of Latin America and the Caribbean's gross forward linkages is, at the aggregate level, relatively low (0.11 in 2010-2011, as measured by the normalized Herfindahl-Hirschman (HH) index).¹⁹ However, other exporting regions show lower concentration levels (see figure 5.A). Within the region, Mexico —highly dependent on the United States market— and, to a lesser extent, Central America and the Caribbean are less diversified than the South American subregion.

Between 1995 and 2011, the concentration of gross forward linkages on the country dimension increased for all Latin American subregions. This is due to the higher participation of Factory Asia (particularly, China) as destination of Latin American and Caribbean countries' exports.²⁰ In fact, for South America, Central America and the Caribbean, gross forward linkages with Factory Asia are significantly more concentrated than those with Factory North America and Factory Europe (see figures 5.B to 5.D). The same holds for other exporting regions.

On the sector dimension, the HH index (normalized by the number of source sectors) shows considerably larger values for Central America and the Caribbean than for Mexico and South America. The concentration level is generally lower for exports to Factory Europe and Factory North America than for

¹⁹ The normalized HH index ranges between 0 and 1. The higher the index, the more concentrated exports are in a few destination markets (an extreme value of 1 would indicate that one market accounts for the country/region's entire forward linkages).

²⁰ The participation of Factory Asia in the region's gross forward linkages doubled between 1995 and 2011 (going from 16% to 33%), while those of Factory Europe and, to a lesser extent, Factory North America fell in this period (from 34% to 20% and from nearly 40% to 36%, respectively).

those oriented to Factory Asia (the main exception is the South American subregion, for which exports to Factory North America are significantly less diversified at the sector level) (see figure 5).





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

Among the six main Latin American suppliers of intermediate inputs for world manufacturing exports, Chile and Costa Rica, along with Mexico, have higher levels of market concentration than Argentina, Brazil and Peru (see figure A.1 in the annex). Chile's forward linkages are highly concentrated in Factory Asia, while those of Mexico and, to a lesser extent, Costa Rica concentrate in Factory North America. On the sector dimension, Costa Rica and Chile show the highest concentration among the six countries, having experienced a large increase between 1995-1996 and 2010-2011 (see figure A.2 in the annex). In the case of Chile, this is explained by the high participation of copper products, for which prices rose substantially in the 2000s. Similarly, Argentina, Brazil and Peru's gross forward linkages — particularly those with Factory Asia— were concentrated in natural resource-based products (like soya, iron, copper and other metals). In contrast, in this period Costa Rica standed out in the region for an increasing concentration of gross forward linkages in high-technology products (particularly, electronic

parts and components).²¹ However, the country has been affected by the closure of Intel's microprocessor assembly plant in 2014, which caused a sharp decline in Costa Rican exports of electronic components.

The analysis of the main imported inputs embodied in Factory North America, Factory Europe and Factory Asia's manufacturing exports shows that they have mostly an intra-regional origin (see figures 6 to 8), particularly in the last two cases (as already shown in diagram 1). It also reveals inter-GVC linkages, mainly in Computer, electronic and optical equipment, where the participation of China has increased considerably. However, it should be pointed out that, in this and other technology-intensive industries (like Electrical machinery and apparatus, n.e.c.), a large proportion of the value added embodied in Chinese gross exports is sourced from abroad, mostly from other Southeast Asian countries, the European Union and the United States.²²

Latin American and Caribbean countries' gross forward linkages with Factory Europe and Factory Asia concentrate in the primary sectors Mining and quarrying and Agriculture (see figures 7 and 8).²³ None of these countries has a significant (i.e., at least 1%) participation in the inputs originated in medium-high or high-technology industries, or even in medium-low-technology industries (the exception is Basic metals, where Brazil and Chile account for a rather significant, though small, share of the total).

For Factory North America, Latin American and Caribbean countries' participation as input suppliers is more diversified in terms of origin sectors. As shown in figure 6, the region sources a significant proportion of inputs from industries like Basic metals (25% of the total), Mining and quarrying (also 25%), Motor vehicles (around 15%), and Electrical machinery and apparatus, n.e.c. (14%). Its share is also relevant in the other main imported inputs embodied in Factory North America's manufacturing exports.²⁴ Although, in most cases, Mexico accounts for the bulk of these inputs, other countries have a significant participation in some sectors (including Chile and Peru in Mining and quarrying and Basic metals, Costa Rica in Computer, electronic and optical equipment, and Brazil in several industries).

²¹ Some Caribbean countries (like Barbados, Saint Kitts and Nevis, and Saint Lucia) also have a large participation of high-technology products in their intermediate exports; however, the share of these countries in the region's forward linkages with Factory North America, Factory Europe and Factory Asia is marginal.

According to the OECD's Trade in Value Added (TiVA) database, in 2011 foreign value added accounted in China for around 55% and 50% of gross intermediate exports from Computer, electronic and optical equipment and Electrical machinery and apparatus n.e.c., respectively. Southeast Asian countries (particularly, Japan, Korea, and Taiwan), the European Union, and the United States were the origin of around 80% and 70% of the total, respectively.

²³ The region's participation as a source of agricultural inputs for Factory Europe, not shown in figure 7, was around 15% (both in 1995 and 2011), with Brazil accounting for over 50% of the total. For Factory Asia, this participation was 22% in 2011 (compared to 5% in 1995), and also concentrated in Brazil (with more than two thirds of the total). Latin American and Caribbean countries' share is also significant in the related industry Food products, beverages and tobacco, with 14% and 19% for Factory Europe and Factory Asia, respectively, in 2011. However, these two sectors represent a very low proportion of the imported inputs embodied in these GVCs' manufacturing exports.

²⁴ Also in this case, Latin America and the Caribbean is an important source of agricultural inputs (24% of the total in 2011), as well as of inputs originated in Food products, beverages and tobacco (19%). The region's participation is also significant in other low-technology industries like Wood and wood products (15%), and Textiles, leather and footwear (10%). These sectors, however, represent a very low share of the imported inputs used in the production of Factory North America's manufacturing exports.



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 2.A). Percentages in brackets indicate the share of each origin country/region in 1995.



Figure 7

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 2.B). Percentages in brackets indicate the share of each origin country/region in 1995.



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 2.C). Percentages in brackets indicate the share of each origin country/region in 1995.

2. Trade agreements and GVC participation

The data presented in this document would indicate that the trade agreements signed by Latin American and Caribbean countries with members of Factory North America, Factory Europe or Factory Asia have not contributed much to promote their forward participation in extra-regional production networks (although causality is not evaluated). The region's linkages with the three major GVCs have remained limited and concentrated in a few origin countries, largely specialized in natural resource-based exports (except for Costa Rica and Mexico).

As shown in table 1, four of the six main Latin American suppliers of foreign intermediates for Factory North America, Factory Europe and Factory Asia (Chile, Costa Rica, Mexico, and Peru) gained preferential access to at least one of the three GVCs in 1995-2011. For Chile, the concentration of vertical specialization-related intermediate exports increased in this period, at least in terms of origin sectors. In contrast, Mexican exports became more diversified in the sector dimension. For Costa Rica and Peru the results were mixed (see figures A.1 and A.2 in the annex).

Table 1
Latin America and the Caribbean: preferential trade agreements with members of GVCs ^a

(In chronological order)

-		Mexico &	
	South America	Central America	The Caribbean
Factory North America Canada	Chile (1997), Peru (2009), Colombia (2011)	Mexico (1994), Costa Rica (2002), Panama (2013), Honduras (2014)	
Mexico	Ecuador (1987 ^b), Peru (1987 ^b , 2012), Paraguay (1994 ^b , 2003 ^b), Colombia (1995), Bolivia (1995, 2010), Chile (1992 ^b , 1999), Argentina (2003 ^b), Brazil (2003 ^b), Uruguay (2003 ^b , 2004)	Cuba (1985 ^b , 2001 ^b), Panama (1986 ^b , 2015), Costa Rica (1995, 2013), Nicaragua (1998, 2012), El Salvador (2001, 2012), Guatemala (2001, 2013), Honduras (2001, 2013)	
United States	Chile (2004), Peru (2009), Colombia (2012)	Mexico (1994), El Salvador (2006), Guatemala (2006), Honduras (2006), Nicaragua (2006), Costa Rica (2009), Panama (2012)	Dominican Republic (2007)
Factory Europe European Union	Chile (2003), Colombia (2013), Peru (2013), Ecuador (2017)	Mexico (2000), Costa Rica (2013), El Salvador (2013), Guatemala (2013), Honduras (2013), Nicaragua (2013), Panama (2013)	CARICOM (2008), Dominican Republic (2008)
Factory Asia Brunei Darussalam China	Chile (2006) Chile (2006), Peru (2010)	Costa Rica (2011)	
Hong Kong	Chile (2014)		

Table 1 (concluded)

		Mexico &	
	South America	Central America	The Caribbean
India	Chile (2007 ^b), Argentina (2009 ^b), Brazil (2009 ^b), Paraguay (2009 ^b), Uruguay (2009 ^b)		
Japan	Chile (2007), Peru (2012)	Mexico (2005)	
Malaysia	Chile (2012)		
Republic of Korea	Chile (2004), Peru (2011), Colombia (2016) Chilo (2006)	Panama (2006), Costa	
Singapore	Peru (2009)	Rica (2013)	
Taiwan		Panama (2004), Guatemala (2006), El Salvador (2008), Honduras (2008), Nicaragua (2008)	
Thailand	Peru (2011), Chile (2015)		
Viet Nam	Chile (2014)		

Source: Author's elaboration on the basis of information from the World Trade Organization's Regional Trade Agreements Information System (RTA-IS) and the Design of Trade Agreements (DESTA) Database (Dür et al., 2014). ^a Numbers within brackets refer to the year in which the agreement entered into force.

^b Partial scope agreement.

For other countries with preferential access to Factory North America, Factory Europe and/or Factory Asia, the share in Latin America and the Caribbean's gross forward linkages remained very low or even decreased. The Dominican Republic, El Salvador, Guatemala, Honduras and Nicaragua accounted jointly for only 3% of the region's gross forward linkages with Factory North America in 2011 (compared with 2.5% in 1995). For these countries, GVC linkages with the United States are mainly of the backward type and concentrate in the labour-intensive assembly segment of the apparel industry. Notwithstanding this, except for Nicaragua, the other four countries diversified their gross forward linkages with Factory North America in the sector dimension (although the concentration in terms of destination markets increased). While El Salvador and Guatemala continued to export largely natural resource-based products, the Dominican Republic and Honduras diversified their exports towards medium-high or high-technology industries.

The Dominican Republic and CARICOM members, who had preferential access to Factory Europe in the studied period, show a low and decreasing participation in the region's total gross forward linkages, as well as mixed results in terms of export concentration. Also, for most Central American countries with some preferential access to Factory Asia, their share in the region's gross forward linkages remained very low (and even decreased); however, exports became more diversified (at least in the sector dimension).

3. Potential further integration into GVCs

As pointed out by Estevadeordal et al. (2013), for countries away from the major GVC regions, the high transport costs implied by long distances are most likely compensated with savings in production costs arising from strong comparative advantages. Thus, Latin America and the Caribbean countries' access to Factory North America, Factory Europe and Factory Asia would take place, at least initially, in sectors where comparative advantages are the strongest.

In order to assess the region's potential as a source of imported intermediates for the three GVCs, a revealed comparative advantage (RCA) index is considered.²⁵ Using data from BACI for 1995-1996 and 2010-2011, an industry-level RCA index is computed for each input-exporting country, considering both total exports and exports to each GVC. This allows distinguishing the industries in which the exporting country is competitive in both the world market and the GVC, from those in which it enjoys a comparative advantage in the GVC despite not being globally competitive.

As shown in table 2, Latin American and Caribbean countries' competitiveness as exporters of intermediate inputs for Factory North America, Factory Europe and Factory Asia is mostly related with natural resource-based industries (many of which are also competitive in the world market).²⁶ Yet, countries like Brazil, Costa Rica, the Dominican Republic, Honduras, Nicaragua and Mexico also show competitiveness in some of the most technology-intensive industries (Computer, electronic and optical equipment; Electrical machinery and apparatus, n.e.c.; or Motor vehicles, depending on the country).²⁷ However, as shown previously in this document, this competitiveness has not translated into a significant participation of these industries in the region's forward linkages with the GVCs.

The overall low forward engagement of Latin American and Caribbean countries in extra-regional value chains could indicate that countries from other regions are more efficient suppliers of the foreign inputs used in the production of the GVCs' manufacturing exports. Although this could be related to structural factors such as countries' distance to the main manufacturing hubs in Factory North America, Factory Europe and Factory Asia, the industries in which many Latin American and Caribbean countries are competitive do not have a significant participation in the imported inputs embodied in those GVCs' exports (see figure 2).

It should be pointed out that, rather than reflect countries' actual comparative advantages, revealed competitiveness might be affected by trade policy, which also plays an active role in promoting countries' engagement in GVCs (Kowalski et al., 2015). Also, some Latin American and Caribbean countries are mainly assemblers or are involved in activities with little domestic transformation. The difference between these countries' gross exports and their value added exports can be large, particularly for technology-intensive products. Thus, the identification of RCAs based on value-added export data would yield very different results.

²⁵ The RCA index, proposed by Balassa (1965), reflects countries' competitiveness in foreign markets by measuring the intensity with which a country exports a product (or group of products), in comparison to the product's share in world exports. A country is considered to have a comparative advantage (disadvantage) in a certain product if the share that the product represents in the country's export basket is higher (lower) than the product's share in world trade. The normalized RCA index is used here. It is computed as: NRCA_{p,t}(i) = $\frac{\text{RCA}_{p,t}(i)-1}{\text{RCA}_{p,t}(i)+1}$, with RCA_{p,t}(i) = $\frac{\text{E}_{p,t}(i)/\text{E}_{p,t}}{\sum_{p} \text{E}_{p,t}(i)/\sum_{p} \text{E}_{p,t}}$ (where the subscript *t* denotes time, $\text{E}_{p,t}(i)$ are country *p*'s exports of intermediate inputs from industry *i*, and $\text{E}_{p,t}$ are country *p*'s total intermediate exports). This index is bounded between -1 and 1, with a positive value indicating that the country has a RCA in industry *i*.

²⁶ Table 2 presents the industries for which each Latin American and Caribbean country had a RCA in 2010-2011 (industries in which the exporting country was only competitive in the GVC, and not in the world market, are indicated in italics). It also shows each industry's participation in countries' vertical specialization-related intermediate exports to the GVC.

²⁷ According to the RCA index, also Antigua and Barbuda, Barbados, Dominica, Saint Kitts and Nevis, Saint Lucia, and Saint Vincent and the Grenadines would be competitive in some medium-high or high-technology industries. It should be noted, however, that Balassa's index is sensitive to the number of exported products. Thus, it takes high values for countries with few exported goods (like these small islands), as the share of each product in total exports is higher than for countries with more diversified export baskets.

Exporting country	Factory North America	Factory Europe	Factory Asia		
South America	South America				
Argentina	Basic metals (75.9%) Textiles (4.6%) Agriculture (2.1%) Food products, beverages and tobacco (1.6%) Mining and quarrying (1.0%) Wood and wood products (0.4%) ^b	Food products, beverages and tobacco (51.3%) <i>Chemical products</i> (20.7%) ^b Textiles (7.0%) Agriculture (5.2%)	Agriculture (49.5%) Food products, beverages and tobacco (27.9%) Textiles (8.0%)		
Bolivia (Plurinational State of)	Basic metals (62.2%) Mining and quarrying (14.6%) Agriculture (6.6%) ^b Wood and wood products (2.0%)	Mining and quarrying (41.7%) Basic metals (22.4%) Food products, beverages and tobacco (12.1%) ^b <i>Textiles (10.9%)^b</i> Agriculture (7.1%) Wood and wood products (3.3%)	Mining and quarrying (74.8%) Wood and wood products (2.0%)		
Brazil	Basic metals (38.8%) Motor vehicles (10.9%) <i>Chemical products</i> (8.5%) ^b Agriculture (6.5%) <i>Rubber and plastics</i> <i>products (4.1%)</i> Pulp, paper and paper products (3.7%) Food products, beverages and tobacco (3.7%) <i>Textiles (3.7%)</i> <i>Other non-metallic mineral</i> <i>products (1.8%)</i> Mining and quarrying (1.5%) ^b Wood and wood products (1.4%)	Agriculture (17.9%) Basic metals (17.0%) Food products, beverages and tobacco (15.9%) Mining and quarrying (12.9%) Pulp, paper and paper products (6.6%) <i>Textiles (4.3%)</i> Wood and wood products (1.3%)	Mining and quarrying (32.6%) Agriculture (20.3%) Pulp, paper and paper products (10.4%) Food products, beverages and tobacco (8.8%)		
Chile	Basic metals (74.2%) Wood and wood products (5.4%) Agriculture (1.9%) Pulp, paper and paper products (1.8%) ^b Food products, beverages and tobacco (0.7%)	Basic metals (74.7%) Mining and quarrying (9.3%) Pulp, paper and paper products (6.2%) Wood and wood products (0.9%) ^b	Basic metals (74.7%) Mining and quarrying (14.4%) Pulp, paper and paper products (7.1%) Wood and wood products (0.7%)		
Colombia	Basic metals (57.2%) ^b Agriculture (22.4%) Mining and quarrying (0.4%)	Agriculture (34.8%) Mining and quarrying (0.1%)	Basic metals (69.2%) ^b Agriculture (8.0%) Mining and quarrying (0.1%) ^b Other non-metallic mineral products (0.0%) ^b		
Ecuador	Agriculture (37.5%) Wood and wood products (9.8%) Mining and quarrying (0.2%)	Agriculture (46.4%) Food products, beverages and tobacco (21.4%) Wood and wood products (5.9%) ^b	Wood and wood products (35.9%) Food products, beverages and tobacco (26.6%) Agriculture (6.1%) ^b Mining and quarrying (0.0%)		

 Table 2

 Latin America and the Caribbean: RCAs in intermediate exports to GVCs, 2010-2011^a

Exporting country	Eactory North America	Eactory Europe	Factory Asia
Paraguay	Agriculture (37.1%)	Agriculture (81.8%)	Agriculture (41.5%)
T uluguuy	Food products, beverages	Food products, beverages	Textiles (40.4%)
	and tobacco (25.6%)	and tobacco (8.0%)	Food products, beverages
	Wood and wood products	Textiles (5.8%)	and tobacco (9.9%) ^o
	(13.070) Textiles (7.4%)		(5.2%)
Peru	Basic metals (89.1%)	Basic metals (44.8%)	Mining and quarrying
	Agriculture (2.7%)	Mining and quarrying	(49.0%)
	Food products, beverages	(29.4%) Agriculture (11.7%)	Basic metals (31.7%)
	Wood and wood products	Food products, beverages	and tobacco (14.8%)
	(0.6%)	and tobacco (7.3%)	Wood and wood products
Uruguov	Chemical products	Pulp, paper and paper	(1.2%) [®]
Oluguay	(35.8%) ^b	products (29.8%) ^b	products (50.2%) ^b
	Textiles (35.0%)	Textiles (29.8%)	Textiles (24.7%)
	Wood and wood products	Wood and wood products	Agriculture (13.1%) ^b
	(10.7%)° Food products beverages	(14.1%)° Food products, beverages	Food products, beverages and tobacco (3.6%)
	and tobacco (1.8%)	and tobacco (8.3%)	Wood and wood products
		Agriculture (6.1%)	(1.7%) ^b
Venezuela (Rolivarian Ropublic	Mining and quarrying	Basic metals (75.7%)	Mining and quarrying
of)	(2.070)	(7.1%)	(29.270)
Mexico and Central A	merica		
Mexico	Motor vehicles (30.2%)	Computer, electronic and	Basic metals (24.4%)
	Electrical machinery and	optical equipment (24.6%)	Mining and quarrying
	Other non-metallic mineral	apparatus, n.e.c. (8.4%) ^b	Electrical machinerv and
	products (0.9%)	Other transport equipment	apparatus, n.e.c. (5.8%) ^b
		(4.6%) ^b Mining and supraving	Motor vehicles (5.0%) ^b
		(1.5%)	
Costa Rica	Computer, electronic and	Computer, electronic and	Computer, electronic and
	optical equipment	optical equipment (76.0%) ^b	optical equipment
	Rubber and plastics	Agriculture (12.8%)	Electrical machinery and
	products (2.0%)		apparatus, n.e.c. (4.0%) ^b
	Agriculture (1.7%)		Agriculture (0.6%)
	and tobacco (0.8%)		
El Salvador	Agriculture (33.7%)	Agriculture (79.2%)	Food products, beverages
	Food products, beverages	Food products, beverages	and tobacco (37.8%)
	Textiles (10.8%)	Rubber and plastics	Agriculture (10.5%)
	()	products (3.6%) ^b	
Guatemala	Agriculture (36.3%)	Agriculture (69.2%)	Food products, beverages
	Easic metals (24.0%) ²	and tobacco (26.5%)	Agriculture (20.8%)
	and tobacco (12.2%)		Wood and wood products
	Mining and quarrying		(1.5%) ^b
	(<i>10.5%)</i> ″ Textiles (6.6%)		
Honduras	Electrical machinery and	Agriculture (87.0%)	Mining and quarrying
	apparatus, n.e.c. (31.5%) ^b	Food products, beverages	(37.5%) ^b
	Agriculture (15.3%)	and tobacco (8.1%) ^b	Agriculture (13.6%)
	and tobacco (5.2%)		

Table 2 (continue)

Exporting country	Factory North America	Factory Europe	Factory Asia
Nicaragua	Basic metals (58.2%) ^b Electrical machinery and apparatus, n.e.c. (18.1%) ^b Agriculture (15.5%) Food products, beverages and tobacco (3.6%)	Agriculture (80.1%) <i>Textiles</i> (9.7%) ^b Food products, beverages and tobacco (8.3%) ^b	Food products, beverages and tobacco (51.9%) Agriculture (12.3%) Wood and wood products (10.2%) ^b
Panama	Chemical products (82.8%) ^b Food products, beverages and tobacco (0.6%)		Chemical products (60.6%) ^b
The Caribbean			
Antigua and Barbuda	Electrical machinery and apparatus, n.e.c. (13.1%) ^b Textiles (9.2%) ^b Fabricated metal products (7.8%) ^b	Agriculture (32.7%)	Machinery and equipment, n.e.c. (35.3%) ^b Basic metals (29.5%) ^b Food products, beverages and tobacco (16.8%) Fabricated metal products (8.9%) ^b
Bahamas	Chemical products (69.8%)		
Barbados	Computer, electronic and optical equipment (55.7%) Food products, beverages and tobacco (9.9%) Fabricated metal products (8.0%) ^b Other transport equipment (6.1%) ^b	Food products, beverages and tobacco (49.5%) <i>Computer, electronic and</i> <i>optical equipment</i> (24.3%) ^b	Fabricated metal products (26.1%) ^b
Belize	Food products, beverages and tobacco (23.7%) <i>Wood and wood products</i> (10.7%) Mining and quarrying (0.0%) ^b	Food products, beverages and tobacco (89.3%)	Agriculture (13.9%) ^b Wood and wood products (11.5%) ^b Food products, beverages and tobacco (10.8%) ^b Textiles (4.1%) ^b
Cuba	Basic metals (36.4%) <i>Chemical products</i> (12.1%) ^b Food products, beverages and tobacco (2.4%)	Basic metals (44.6%) Food products, beverages and tobacco (24.9%)	Basic metals (77.4%) ^b Food products, beverages and tobacco (19.9%)
Dominica	Chemical products (39.3%) Electrical machinery and apparatus, n.e.c. (17.4%) ^b Fabricated metal products (7.3%) ^b Pulp, paper and paper products (2.4%) ^b	Mining and quarrying (19.7%) ^b Food products, beverages and tobacco (13.3%) ^b Agriculture (8.9%) ^b	Basic metals (25.5%) Machinery and equipment, n.e.c. (17.4%) ^b Electrical machinery and apparatus, n.e.c. (15.9%) ^b
Dominican Republic	Electrical machinery and apparatus, n.e.c. (25.6%) Textiles (7.9%) ^b Food products, beverages and tobacco (6.2%) Agriculture (5.5%)	Agriculture (38.3%) Electrical machinery and apparatus, n.e.c. (14.1%) ^b	Basic metals (53.1%)
Grenada	Agriculture (14.1%)	Agriculture (54.1%)	Chemical products (12.3%) Textiles (4.3%) Motor vehicles (0.4%) ^b

Table 2 (continue)

Exporting country	Factory North America	Factory Europe	Factory Asia
Guyana	Basic metals (97.1%) Food products, beverages and tobacco (0.9%) Wood and wood products (0.3%)	Food products, beverages and tobacco (77.4%) Mining and quarrying (15.8%) Wood and wood products (1.9%)	Agriculture (35.5%) Wood and wood products (17.9%) Food products, beverages and tobacco (0.9%) ^b
Haiti		Chemical products (61.9%) ^b Agriculture (28.3%) Food products, beverages and tobacco (4.2%)	Textiles (1.8%) ^b
Jamaica	Basic metals (85.7%) Food products, beverages and tobacco (4.4%) Mining and quarrying (4.0%)	Basic metals (73.7%) Food products, beverages and tobacco (19.9%)	Food products, beverages and tobacco (13.0%) ^b Agriculture (6.9%)
Saint Kitts and Nevis	Computer, electronic and optical equipment (74.8%) ^b Electrical machinery and apparatus, n.e.c. (15.0%) Pulp, paper and paper products (3.3%) ^b	Electrical machinery and apparatus, n.e.c. (71.4%) ^b Machinery and equipment, n.e.c. (16.9%) ^b	Computer, electronic and optical equipment (81.5%) ^b Fabricated metal products (1.8%) ^b Agriculture (0.5%)
Saint Lucia	Computer, electronic and optical equipment (87.8%) Electrical machinery and apparatus, n.e.c. (4.7%) ^b		Computer, electronic and optical equipment (36.5%) ^b Machinery and equipment, n.e.c.(3.1%) ^b Food products, beverages and tobacco (1.8%) Electrical machinery and apparatus, n.e.c.(1.3%) ^b
Saint Vincent and the Grenadines	Basic metals (34.3%) ^b Other transport equipment (11.7%) Food products, beverages and tobacco (3.4%)		Fabricated metal products (29.8%) ^b Electrical machinery and apparatus, n.e.c.(16.6%) ^b
Suriname	Basic metals (98.2%)	Basic metals (91.0%) Food products, beverages and tobacco (3.0%) Wood and wood products (0.5%)	Agriculture (28.3%) Wood and wood products (17.8%) ^b Food products, beverages and tobacco (1.5%) ^b
Trinidad and Tobago	Chemical products (70.7%) Basic metals (27.0%) Mining and quarrying (0.9%) Food products, beverages and tobacco (0.8%)	Basic metals (50.0%) ^b Chemical products (46.5%) Mining and quarrying (0.0%) ^b	Mining and quarrying (7.1%) ^b

Table 2 (concluded)

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

^a Industries are ranked, from highest to lowest, on the basis of their share in countries' vertical specialization-related intermediate exports to the GVC (indicated in brackets). In italics, industries with RCA in the GVC but not in the world market.

^b Industries that were not competitive in 1995-1996.

III. Intra-regional production networks in Latin America

In contrast to the strong role of regional value chains in North America, Europe and Southeast Asia, countries in Latin America and the Caribbean have weak intra-regional production linkages (OECD, 2015). Most foreign intermediate inputs embodied in Argentina, Brazil, Chile, Colombia, Costa Rica, Mexico and Peru's manufacturing exports are imported from outside the region (particularly, those with a higher technology content). The participation of China increased significantly since the early 2000s, in detriment of other extra-regional origins (the United States and the European Union).

Notwithstanding the above general pattern, the seven Latin American countries for which data are available exhibit significant differences in the composition of their backward linkages. The analysis carrried out in this section characterizes each country's backward participation in international value chains and explores the potential for deeper intra-regional production integration.

A. Current linkages

1. Argentina

The high natural resource intensity of Argentina's manufacturing exports reflects in a relatively low level of vertical specialization (18% of gross exports in 2011). However, as shown in figure 9, there are large differences across industries. Among the main exporting sectors, the share of foreign intermediates in gross exports ranged in 2011 between 11% for Food products, beverages and tobacco (which accounted for almost half of manufacturing exports) and 33% for the more technology intensive industry Motor vehicles (18% of total exports). Despite these differences, the use of foreign inputs increased significantly for all manufacturing sectors with respect to 1995.

Argentina stands out in the region for the high level of backward integration with its MERCOSUR partner Brazil, the main origin of imported intermediate goods in Argentine manufacturing exports (almost 40% of the total in 2011, compared to nearly 30% in 1995) (see figure 10.A). Gross backward linkages with other Latin American and Caribbean countries represented only 6.5% of the total in 2011 (compared to 9.3% in 1995), and were concentrated in Chile (2.6%), Mexico (1.8%) and Uruguay (1.0%) (another MERCOSUR partner). The European Union, the United States, China and other Southeast Asian countries

(particularly, Japan, Korea, Thailand, and India) were, along with Brazil, the main suppliers of foreign intermediates for Argentina's manufacturing exports.²⁸



(Percentages of gross exports)





^a Percentages in brackets indicate the share of each industry in total manufacturing exports in 2011.

The composition by source industry shows that the main imported intermediate goods in Argentine gross exports corresponded in 2011 to Chemical products and Motor vehicles, followed by Basic metals, Fabricated metal products, Machinery and equipment, n.e.c., and Rubber and plastics products (see figure 10.B). Brazil is the main or one of the main suppliers of most of these inputs, with a generally increasing participation (see figure A.3 in the annex). The share of Brazil is particularly high in the case of Motor vehicles, Basic metals, and Rubber and plastics products, as well as in other intermediates with a lower participation in Argentina's input basket (like Pulp, paper and paper products, and Electrical machinery and apparatus, n.e.c.). Brazil is also the main source of mining inputs (particularly, iron ores and concentrates).²⁹

The competition of China has affected the United States and the European Union, who lost participation in Argentina's gross backward linkages since the 2000s. Despite representing a relatively low (and decreasing) share of total inputs, it is worth mentioning the most technology-intensive sectors Computer, electronic and optical equipment and Electrical machinery and apparatus, n.e.c., where the participation of China reached the highest levels (affecting also other Southeast Asian countries). However, as pointed out in section II, in these sectors a large proportion of the value added embodied in

²⁸ In addition to Brazil and the rest of MERCOSUR's founding members, several Latin American countries benefited from some degree of preferential access to the Argentine market in the studied period (including Chile, Mexico, and the Andean Community's members) (see table A.3 in the annex).

²⁹ Brazil also has a large participation in Agriculture and all low-technology industries (see table A.4 in the annex).

Chinese gross exports is sourced from abroad, mostly from other Southeast Asian countries, the European Union and the United States.

At the aggregate level, Latin American and Caribbean countries (excluding Brazil) had a larger participation in low and medium-low-technology industries (particularly, Wood and wood products, Pulp, paper and paper products, Food products, beverages and tobacco, Other non-metallic mineral products, Rubber and plastics products, and Basic metals), which account in general for a small share of Argentina's gross backward linkages. However, along with Brazil, a few countries had rather significant participations in more technology-intensive industries, like Mexico in Computer, electronic and optical equipment, Motor vehicles, and Electrical machinery and apparatus, n.e.c. (see table A.4 in the annex for details on each industry's main intra-regional origin countries).



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

^a Percentages in brackets indicate the share of each origin country/region (figure 10.A) or industry (figure 10.B) in 1995.

2. Brazil

Brazil's level of vertical specialization is lower than those of the other Latin American countries considered (14% of gross exports in 2011).³⁰ This would mainly be related with the country's larger market size, rather than with its export specialization, since all manufacturing sectors show lower backward linkages than in the other six countries (with a few exceptions). However, for many industries the level of vertical specialization is lower in Brazil than in other large countries like France, Germany, the United Kingdom or the United States, which would be explained by these countries' higher level of development.³¹

Among Brazil's main exporting industries, the share of foreign intermediates in gross exports ranged in 2011 between 10% for Food products, beverages and tobacco (the largest exporter, with around 30% of the total) and 20% for Motor vehicles (10% of total exports) (see figure 11). To a lesser extent than in the case of Argentina, the use of imported inputs also increased in Brazil for all manufacturing sectors between 1995 and 2011.

³⁰ Brazil shows the lowest level of vertical specialization among the 63 countries included in the OECD's ICIO tables.

³¹ A larger domestic market is associated with a lower backward participation in GVCs, due to countries' larger capabilities for sourcing intermediates domestically. However, developed countries tend to source more from abroad (Kowalski et al., 2015).

Figure 11 Brazil: imported intermediates embodied in gross manufacturing exports by exporting industry, 1995 and 2011ª



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables.

^a Percentages in brackets indicate the share of each industry in total manufacturing exports in 2011.

Brazil is also characterized by a lower level of intra-regional production integration than the other Latin American countries (the only exception is Mexico, when considered as part of the Latin American region³²). In 2011, less than 20% of the imported intermediate goods embodied in Brazilian manufacturing exports were sourced from the region (the same proportion as in 1995) (see figure 12.A). Argentina (6.2%) —Brazil's largest MERCOSUR partner—, Chile (5.4%), Mexico (1.9%) and Peru (1.5%) accounted for the bulk of these intra-regional inputs. The other two founding members of MERCOSUR, Paraguay and Uruguay, represented less than 10% of the intra-regional total (with a joint participation of only 1.6% in total foreign intermediates). Most imported inputs in Brazil's manufacturing exports came from the European Union, the United States, China and other Southeast Asian countries (particularly, Japan and Korea).³³

The main origin industries of the imported intermediate goods embodied in 2011 in Brazil's gross manufacturing exports were Chemical products, Basic metals, and Machinery and equipment, n.e.c. (see figure 12.B). Other industries with a significant participation include Other transport equipment, Computer, electronic and optical equipment, Rubber and plastics products, and Fabricated metal products. The European Union, the United States, China and the rest of South and East Asia accounted for the majority of all these inputs, with particularly high shares in the most technology-intensive ones (see figure A.4 in the annex). Also in this case, the gross participation of China increased significantly in the studied period (reaching higher levels in Computer, electronic and optical equipment, and Electrical machinery and apparatus, n.e.c.), in detriment of the United States and the European Union.

Intra-regional inputs had a larger participation in the medium-low-technology industries Basic metals (almost 40% of the total) and Rubber and plastics products (22%) (see figure A.4 in the annex). Their share was also significant in Motor vehicles (16%); however, Argentina —the main input supplier in this sector within the region—lost almost 20 percentage points between 1995 and 2011, due to the competition of South and East Asia. As for other intermediates with a lower participation in Brazil's gross backward linkages,

³² Although Mexico is a Latin American country, it is much more economically integrated with North America (particularly, with the United States).

³³ The founding members of MERCOSUR, Chile, Mexico, the Andean Community's members, and some Caribbean countries enjoyed, in different degrees, preferential access to the Brazilian market during the studied period (see table A.3 in the annex).

Latin American and Caribbean countries accounted for a large proportion of those originated in the primary sectors Mining and quarrying and Agriculture (over 50% of the total), and in the low-technology industries Wood and wood products (48%) and Food products, beverages and tobacco (36%) (for details, see table A.4 in the annex).

Figure 12 Brazil: composition of imported intermediate goods embodied in gross manufacturing exports, 1995 and 2011^a



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Percentages in brackets indicate the share of each origin country/region (figure 12.A) or industry (figure 12.B) in 1995.

3. Chile

The level of Chile's gross backward linkages (22% of gross exports in 2011) is comparable to those of Colombia (20%) and Peru (19%), two countries with a similar export specialization pattern.³⁴ As shown in figure 13, the three main Chilean exporting sectors in manufacturing, the natural resource-based Basic metals, Food products, beverages and tobacco and Pulp, paper and paper products, have similar vertical specialization levels (around 20%). For most manufacturing sectors, the use of foreign inputs increased between 1995 and 2011 (the exceptions are the technology-intensive industries Machinery and equipment, n.e.c., Computer, electronic and optical equipment and Motor vehicles, which account for a low share of total exports).

Latin American countries are an important source of foreign inputs for Chile's manufacturing exports, with a joint participation of nearly 40% in 2011 (compared to around 35% in 1995) (see figure 14.A). These intra-regional linkages concentrate in Chile's Pacific Alliance partners Peru (11.7%) and Mexico (3.7%), and the two largest MERCOSUR members (Brazil and Argentina, with 8.3% and 7.9%, respectively). The United States, the European Union, China and the rest of South and East Asia (particularly, Korea, Japan, Taiwan, and India) account for the bulk of extra-regional inputs.³⁵

³⁴ Chile's gross vertical specialization level is also similar to those of Australia (25%) and New Zealand (21%), two extra-regional natural resource-intensive economies.

³⁵ As shown in table A.3 in the annex, Mexico, Peru, the United States, the European Union, China and several other Southeast Asian countries enjoyed preferential access to the Chilean market during the studied period. Also, Chile had FTAs or partial scope agreements in place with other extra-regional countries, as well as with members of the Andean Community, CARICOM and MERCOSUR.


Figure 13 Chile: imported intermediates embodied in gross manufacturing exports by exporting industry,

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables.

^a Percentages in brackets indicate the share of each industry in total manufacturing exports in 2011.



A. By geographical origin B. By source industry^b Pulp, pape and pape products 3.8 Textiles Fabricated netal products Centra The Caribbear leathe Rest of South rica 0.4 (0.2 est of the Agriculture 3.8 Other Americ 3.8 (4.4) 2.1 3.1 (4.4) 5.8 (7.3) (0.8 **3.1** (2.4) dustries (4.7) (8.6) 5.5 (10.1)United States Mexico 3.7 (4.6) Food products 16 7 sic metals verages and tobacco (26.9) **18.3** (18.4) 4.6 (5.9) Peru Europear Union Manufacturing nec; recycling 17.1 (0.7) 11.7 Machinery and **14.7** (24.7) (2.1) equipment, neo 6.3 Brazi (13.0) 8.3 (11.8) Rubber and China **17.1** (1.0) products Argentin 7.9 Mining and Rest of South and East Asia quarrying 15.3 (4.2) 6.5 (5.2) products (9.5) **12.6** (22.4) 7.7 (6.4)

Chile: composition of imported intermediate goods embodied in gross manufacturing exports,

(Percentages of the total)

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

^a Percentages in brackets indicate the share of each origin country/region (figure 14.A) or industry (figure 14.B) in 1995.

^b Manufacturing n.e.c.; recycling corresponds mostly to metal waste and scrap.

Most imported intermediate goods used in the production of Chilean manufacturing exports originate in mining-related sectors (see figure 14.B). Latin American and Caribbean countries (particularly, Peru) are the main suppliers of inputs from Mining and quarrying, while inputs from Basic metals (as well as metal waste) are sourced mostly from outside the region (mainly, from South and East

Asia and the European Union) (see table A.4 and figure A.5 in the annex). Extra-regional countries are also the main origin of most of the other imported intermediate goods embodied in Chile's manufacturing exports (including, Chemical products, Rubber and plastics products, and Machinery and equipment, n.e.c.), with an increasing participation of China. The exceptions are Food products, beverages and tobacco and Agriculture, where Latin American and Caribbean countries have a larger participation (for details, see table A.4 in the annex).

4. Colombia

The aggregate level of Colombia's gross vertical specialization (20% of gross exports in 2011) is driven mainly by Basic metals, Chemical products and Food products, beverages and tobacco, the country's three largest exporting sectors in manufacturing (see figure 15). While in Basic metals the use of foreign inputs fell by one third between 1995 and 2011, Chemical products, Food products, beverages and tobacco and most of the other manufacturing sectors experienced an increase in this period. Also, unlike the case of Chile, in Colombia vertical specialization is significantly higher in the most technology intensive industries (particularly, Electrical machinery and apparatus, n.e.c., Motor vehicles, and Other transport equipment), which jointly account for a rather important share of total manufacturing exports.

Figure 15 Colombia: imported intermediates embodied in gross manufacturing exports by exporting industry, 1995 and 2011^a



(Percentages of gross exports)

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables.

^a Percentages in brackets indicate the share of each industry in total manufacturing exports in 2011.

In 2011, nearly 30% of the foreign intermediates used in the production of Colombia's manufacturing exports was imported from the Latin American region (somewhat more than in 1995), with Mexico (8.3%), Brazil (6.8%), Peru (2.8%), Argentina, Chile and Venezuela (2.3% each), and Ecuador (1.2%) as the main origin countries (see figure 16.A). Also in this case, the United States, the European

Union, China and the rest of South and East Asia (particularly, Japan, Korea, India, and Taiwan) accounted for the bulk of extra-regional inputs.³⁶

The composition by source industry shows that the main imported intermediates in Colombia's gross exports corresponded in 2011 to Chemical products and Basic metals (see figure 16.B). The mediumhigh or high-technology industries Other transport equipment, Machinery and equipment, n.e.c., Electrical machinery and apparatus, n.e.c., Motor vehicles and Computer, electronic and optical equipment, sourced mainly from extra-regional countries, also had a significant participation.





^a Percentages in brackets indicate the share of each origin country/region (figure 16 A) or industry (figure 16 B) in 1995.

Latin American and Caribbean countries accounted for a larger proportion of the inputs originated in Basic metals, Pulp, paper and paper products and Rubber and plastics products, although their share in Electrical machinery and apparatus, n.e.c., Chemical products and Motor vehicles was also important (see figure A.6 in the annex). Notwithstanding this, the region sourced a larger proportion of other intermediates with a lower participation in Colombia's input basket (particularly, those originated in Agriculture and the low-technology industries Food products, beverages and tobacco and Wood and wood products) (for details, see table A.4 in the annex).

5. Costa Rica

Costa Rica has, after Mexico, the second highest gross vertical specialization level in Latin America (7 countries) (39% in 2011). Both countries specialize in assembling and processing foreign intermediates, which are subsequently exported. Unlike other countries in the region, Costa Rican manufacturing exports concentrate in the high-technology sector Computer, electronic and optical equipment (65% of the total in 2011), where the share of foreign inputs in gross exports reached 45% (see figure 17).³⁷ In contrast, the

³⁶ Several Latin American and Caribbean countries had some degree of preferential access to the Colombian market in the studied period, including the members of the Andean Community, CACM, MERCOSUR and CARICOM, Chile, and Mexico. Colombia has also signed several extra-regional FTAs, including those with the United States and the European Union (entered into force after 2011) (see table A.3 in the annex).

³⁷ As noted before, Costa Rica's exports of high-technology products were significantly affected by the closure of Intel's microprocessor assembly plant in 2014.

natural resource-based industry Food products, beverages and tobacco (around 14% of gross manufacturing exports) had a vertical specialization level of 20%. In these two sectors, the use of foreign inputs increased between 1995 and 2011 (particularly, in Computer, electronic and optical equipment), but most of the other industries experienced a decline in this period.





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables.

^a Percentages in brackets indicate the share of each industry in total manufacturing exports in 2011.

The United States concentrate more than half of Costa Rica's gross backward linkages (see figure 18.A). However, Latin American countries have an important, though decreasing, participation (21% of the total in 2011, compared to 27% in 1995). Mexico (5.1%), Chile (2.5%), Brazil (2.3%), Colombia (1.5%) and Panama (1.4%), along with Costa Rica's CACM partners Guatemala (2.4%), El Salvador (1.4%) and Honduras (1.0%), account for the bulk of these intra-regional linkages. The share of China and, to a lesser extent, other Southeast Asian countries (particularly, Japan) in total foreign inputs has increased, although it is relatively low compared to Asia's participation in other Latin American countries' backward linkages. The European Union's share, in contrast, fell significantly in the studied period.³⁸

A large proportion of the foreign inputs embodied in Costa Rica's manufacturing exports originates in the high-technology industry Computer, electronic and optical equipment (see figure 18.B). Other sectors with a significant participation include Chemical products and the medium-low-technology industries Basic metals, Fabricated metal products and Rubber and plastics products. The main origin of these inputs (particularly, those from Computer, electronic and optical equipment), as well as other intermediates with a lower participation in Costa Rica's input basket, are extra-regional countries (see figure A.7 and table A.4 in the annex). The Latin American region accounts for a larger proportion of inputs from Mining and quarrying, and low and medium-low-technology industries (among the main source sectors, Basic metals and Rubber and plastics products). A few Latin American countries also have a significant participation in medium-high or high-technology industries (particularly, Brazil and Mexico).

³⁸ In addition to the CACM's partners (El Salvador, Guatemala, Honduras and Nicaragua), CARICOM countries, Chile, Colombia, the Dominican Republic, Mexico and Panama benefited from some preferential access to the Costa Rican market during the studied period. Also, before 2011 Costa Rica put in place FTAs with Canada and the United States (see table A.3 in the annex).



Figure 18 Costa Rica: composition of imported intermediate goods embodied in gross manufacturing exports, 1995 and 2011^a

(Percentages of the total)

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Percentages in brackets indicate the share of each origin country/region (figure 18.A) or industry (figure 18.B) in 1995.

6. Mexico

Integrated into Factory North America through NAFTA, Mexico has a relatively high level of vertical specialization (44% of gross exports in 2011). Among Mexico's main exporting industries, gross backward linkages ranged in 2011 between 17% for Basic metals and 64% for Computer, electronic and optical equipment (see figure 19).³⁹ Except for Basic metals, the use of foreign inputs increased for all Mexican manufacturing sectors between 1995 and 2011.

The United States remain as the main origin of the imported intermediates used in the production of Mexico's manufacturing exports, but China and other Southeast Asian countries (particularly, Japan, Korea, Taiwan, and Malaysia) have accounted for an increasing share of the total (see figure 20.A). Thus, between 1995 and 2011 the United States' participation fell from nearly 80% to around 50%, while that of Asian countries increased from 10% to 35%.⁴⁰ Mexico's backward linkages with Latin American and Caribbean countries are very low (around 4% of the total in 2011).⁴¹

Most of the imported intermediate goods embodied in Mexico's manufacturing exports originate in medium-high or high-technology industries (particularly, Computer, electronic and optical equipment, Machinery and equipment, n.e.c., Electrical machinery and apparatus, n.e.c., and Motor vehicles) (see figure 20.B). Some medium-low-technology industries (Basic metals, Fabricated metal products, and Rubber and plastics products) also have a significant participation. Except for Computer, electronic and optical equipment, where China and the rest of South and East Asia account for three quarters of the total, the United States is still the main supplier of all these inputs (although the competition of Asian countries has particularly affected American exports) (see figure A.8 in the annex).

³⁹ The OECD's ICIO tables include a breakdown for Mexico between global manufacturers and other enterprises. For global manufacturers, the aggregate level of vertical specialization was nearly 60% in 2011 (reaching 76% for Computer, electronic and optical equipment). In contrast, backward linkages for non-global manufacturers represented only 20% of gross exports.

⁴⁰ China's participation went in this period from less than 1% to 16%. However, as noted previously in this document, in value-added terms this participation would be considerably lower (particularly, in some of the most technology-intensive sectors).

⁴¹ During the studied period, several Latin American countries enjoyed some degree of preferential access to the Mexican market, including the members of the Andean Community, CACM and MERCOSUR, and Chile. Also, Mexico put in place several extra-regional FTAs, including those with the European Union and Japan (see table A.3 in the annex).



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables.

^a Percentages in brackets indicate the share of each industry in total manufacturing exports in 2011.



(Percentages of the total) By geographical origin B. By source industry Α. Rest of the World Rubber and Central The Caribbean Textiles Pulp, paper and paper America 0.1 plastics leather and 1.1 (0.1) (0.2) 3.9 (2.5) prod 2.3 (4.2) products footwear Other 2.0 (5.1) 6 5 ndustries (9.5) 5.3 Rest of South (4.5) Americ 1.2 Chemica products (1.1) 8.1 Computer Brazi 1.2 (7.5) electronic and optical equip. 26.1 (23.9) (1.0) Fabricated metal products 8.5 Rest of South and East Asia 19.1 United States (7.3) 48.5 Machinery and equipment, nec 11.9 (9.8) (76.6) Motor vehi 9.2 (5.8) (5.4)China 15.8 Electrica Basic metals machinery and apparatus, nec 9.6 (15.3) (0.4) European
 Union **10.4** (11.4) 9.1 (8.2)

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Percentages in brackets indicate the share of each origin country/region (figure 20.A) or industry (figure 20.B) in 1995. The composition of Mexico's backward linkages with Latin American and Caribbean countries by source industry reveals a poor intra-regional production integration. These countries have a larger participation in Mining and quarrying and the low-technology industries Wood and wood products and Food products, beverages and tobacco, which account for a very low share of the foreign intermediates embodied in Mexican manufacturing exports. Only Brazil and Costa Rica have a somewhat significant participation in medium-high or high-technology industries (see table A.4 in the annex).

7. Peru

With a similar export specialization pattern than Chile and Colombia, Peru's aggregate level of gross backward linkages (19% of gross exports in 2011) is given mainly by the natural resource-based industries Basic metals and Food products, beverages and tobacco (which accounted jointly for more than 70% of manufacturing exports) (see figure 21). For these and most other Peruvian manufacturing sectors, the use of foreign inputs increased between 1995 and 2011.





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables.

^a Percentages in brackets indicate the share of each industry in total manufacturing exports in 2011.

The United States, the European Union and, increasingly, China and the rest of South and East Asia (particularly, Korea, India, Japan, and Taiwan) account for the majority of the imported inputs used in the production of Peru's manufacturing exports (see figure 22.A). However, Latin American countries are an important, though decreasing, source of foreign intermediates, with a joint participation of around 30% in 2011 (compared to nearly 40% in 1995). Like in the case of Chile, Peru's intra-regional linkages concentrate in the two largest MERCOSUR members (Brazil and Argentina, with 7.7% and 7.6% of the total, respectively), and the Pacific Alliance partners Chile, Mexico and Colombia (4.5%, 3.3% and 2.6% of the total, respectively).⁴²

⁴² Along with the Andean Community members, Chile, Mexico and MERCOSUR countries benefited from some preferential access to the Peruvian market during the studied period. In addition, by the end of the period Peru implemented FTAs with Canada, the United States, China and other Asian countries (see table A.3 in the annex).

In terms of source sectors, Peru's backward linkages concentrate in Basic metals and Chemical products (see figure 22.B). The other main origins are mostly primary sectors and low-technology industries (Textiles, leather and footwear, Pulp, paper and paper products, and Food products, beverages and tobacco). Most of non-primary inputs are sourced mainly from outside the region (see figure A.9 in the annex). Latin American and Caribbean countries account for a larger proportion of the inputs originated in the primary sectors (Mining and quarrying, and Agriculture), Food products, beverages and tobacco, and other low and medium-low-technology industries with a lower participation in Peru's backward linkages. However, countries like Argentina, Brazil, Chile, Colombia, Costa Rica and Mexico also have a significant participation in medium-high or high-technology industries (for details, see table A.4 in the annex).

Figure 22 Peru: composition of imported intermediate goods embodied in gross manufacturing exports, 1995 and 2011ª



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

^a Percentages in brackets indicate the share of each origin country/region (figure 22.A) or industry (figure 22.B) in 1995.

B. Intra-regional export potential

According to the RCA index, Latin American and Caribbean countries' intra-regional competitiveness in intermediate goods concentrates in primary sectors and low and medium-low-technology industries (see table 3). However, countries like Argentina, Brazil, Costa Rica, the Dominican Republic, Honduras, Mexico, Nicaragua, Panama and Uruguay also show competitiveness in more technology-intensive sectors (Machinery and equipment, n.e.c., Electrical machinery and apparatus, n.e.c., Computer, electronic and optical equipment, Motor vehicles, or Other transport equipment, depending on the country).⁴³ Yet, as in the case of Factory North America, Factory Europe and Factory Asia, the share of these sectors in the region's gross forward linkages with Argentina, Brazil, Chile, Colombia, Costa Rica, Mexico and Peru is relatively low. Moreover, regardless of their technology intensity, Latin American and Caribbean industries with revealed intra-regional competitiveness do not necessarily account for a significant portion of the imported inputs embodied in the region's manufacturing exports (as shown previously in this section).

For many countries, comparative advantages in some industries —particularly, those with higher technology content— are limited to the regional level. For instance, Uruguay, a country with preferential access and geographical proximity to Argentina, enjoys in that market comparative advantages in two medium-high or high-technology industries (Motor vehicles and Electrical machinery and apparatus, n.e.c.) that are not competitive in any of the major GVCs or the world market. Similarly, preferential access would contribute to explain Argentina's comparative advantage in the Brazilian and Mexican markets in Motor vehicles (geographical proximity to Brazil would play a role as well).⁴⁴ Also, Chile and Uruguay —who enjoy different degrees of preferential access to the Colombian market— show competitiveness in Colombia in some technology-intensive industries that are not competitively exported to any of the major GVCs or the world market. The same would hold for other countries (including Brazil and Mexico), and for sectors with lower technology content. This reveals the importance of regional integration for Latin American and Caribbean countries, particularly the smallest ones, to build additional export capacities.

⁴³ As pointed out in section II, results on revealed competitiveness of the small Caribbean countries should be interpreted with caution, bearing in mind the limitations of the Balassa index. However, the data presented in table 3 shows that, for many Caribbean countries, intra-regional gross forward linkages concentrate in technology-intensive industries (although these countries' participation as intermediate input suppliers for Argentina, Brazil, Chile, Colombia, Costa Rica, Mexico and Peru is marginal).

⁴⁴ In the case of Mexico, Argentina did not have revealed competitiveness in Motor vehicles in 1995-1996, when the automotive agreement had not yet been implemented (see table 3).

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
South America							
Argentina		Agriculture (30.0%) Rubber and plastics products (9.0%) ^b Food products, beverages and tobacco (8.6%) Motor vehicles (7.7%) Pulp, paper and paper products (3.1%) ^b Wood and wood products (0.9%)	Agriculture (18.3%) Food products, beverages and tobacco (13.7%) Chemical products (12.9%) ^b Mining and quarrying (11.2%) Pulp, paper and paper products (6.5%) ^b Other non-metallic mineral products (1.4%)	Agriculture (40.2%) Food products, beverages and tobacco (32.8%)	Textiles (29.7%) ^b Wood and wood products (0.3%)	Basic metals (35.7%) Motor vehicles (16.7%) ^b Textiles (11.6%) Mining and quarrying (6.6%) ^b Food products, beverages and tobacco (1.8%)	Agriculture (34.2%) Food products, beverages and tobacco (19.1%) Chemical products (16.2%) ^b Pulp, paper and paper products (3.5%) ^b Wood and wood products (0.3%) ^b
Bolivia (Plurinational State of)	Mining and quarrying (55.0%) Agriculture (19.4%) ^b Wood and wood products (7.6%)	Mining and quarrying (47.7%) Wood and wood products (2.7%)	Food products, beverages and tobacco (26.8%) Basic metals (6.9%) Mining and quarrying (3.2%) ^b Wood and wood products (2.8%) <i>Textiles (2.1%)^b</i> Agriculture (2.0%) <i>Other non-metallic</i> <i>mineral products</i> (0.3%) ^b	Food products, beverages and tobacco (87.4%) Mining and quarrying (5.0%) ^b	Food products, beverages and tobacco (55.5%) ^b Mining and quarrying (10.2%) ^b Machinery and equipment, n.e.c. (6.2%) ^b Other non-metallic mineral products (3.4%) ^b	Basic metals (63.8%) ^b Mining and quarrying (16.1%) Wood and wood products (7.2%) <i>Textiles (2.8%)^b</i> Food products, beverages and tobacco (0.6%) ^b	Mining and quarrying (48.3%) Food products, beverages and tobacco (38.0%) Agriculture (8.4%)

 Table 3

 Latin America and the Caribbean: RCAs in intra-regional intermediate exports, 2010-2011^a

Table 3	(continued)
---------	-------------

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Brazil	Motor vehicles (30.6%) Mining and quarrying (13.5%) Basic metals (13.0%) <i>Rubber and plastics</i> <i>products</i> (6.7%) Pulp, paper and paper products (3.6%) Food products, beverages and tobacco (1.6%) Other non-metallic mineral products (0.7%)		Basic metals (30.2%) <i>Rubber and plastics</i> <i>products (8.1%)</i> Pulp, paper and paper products (6.3%) Motor vehicles (3.0%) Mining and quarrying (0.6%) ^b	Basic metals (28.3%) Chemical products (23.8%) ^b Motor vehicles (11.0%) Rubber and plastics products (5.8%) Electrical machinery and apparatus, n.e.c. (5.1%) Mining and quarrying (4.6%) ^b Pulp, paper and paper products (4.5%) ^b Agriculture (3.7%) ^b Other non-metallic mineral products (0.7%)	Basic metals (50.1%) Fabricated metal products (12.5%) Rubber and plastics products (9.6%) ^b Chemical products (9.6%) ^b Textiles (5.1%) ^b Machinery and equipment, n.e.c. (3.4%) Motor vehicles (2.6%) Food products, beverages and tobacco (0.7%) ^b Other non-metallic mineral products (0.4%)	Motor vehicles (23.3%) Basic metals (21.5%) Machinery and equipment, n.e.c. (14.9%) Rubber and plastics products (8.4%) ^b Textiles (4.1%) Mining and quarrying (3.3%) Other non-metallic mineral products (1.9%) Other transport equipment (0.7%) Wood and wood products (0.4%)	Basic metals (59.4%) Pulp, paper and paper products (5.8%) <i>Rubber and plastics</i> <i>products (4.0%)</i> Motor vehicles (1.7%) <i>Other non-metallic</i> <i>mineral products</i> (0.2%)
Chile	Fabricated metal products (20.1%) Basic metals (19.9%) Pulp, paper and paper products (16.7%) Rubber and plastics products (12.3%) ^b Food products, beverages and tobacco (1.7%) Agriculture (1.7%) Other non-metallic mineral products (1.5%) ^b Textiles (1.1%) ^b Wood and wood products (0.9%)	Basic metals (61.5%) Mining and quarrying (25.9%) <i>Fabricated metal</i> <i>products</i> (2.6%) ^b Pulp, paper and paper products (1.0%)		Basic metals (28.3%) Pulp, paper and paper products (23.0%) Electrical machinery and apparatus, n.e.c. (7.0%) ^b Rubber and plastics products (6.8%) ^b Wood and wood products (5.3%) Fabricated metal products (4.9%) ^b Motor vehicles (4.6%) ^b Food products, beverages and tobacco (2.7%)	Basic metals (78.1%) <i>Rubber and plastics</i> <i>products</i> (12.3%) ^b Pulp, paper and paper products (3.5%) ^b Wood and wood products (0.5%) <i>Food products</i> , <i>beverages and</i> <i>tobacco (0.5%)^b</i>	Basic metals (51.0%) Wood and wood products (10.6%) Pulp, paper and paper products (5.3%) ^b Food products, beverages and tobacco (0.5%)	Pulp, paper and paper products (18.1%) <i>Rubber and plastics</i> products (6.6%) Wood and wood products (4.2%) <i>Food products,</i> <i>beverages and</i> <i>tobacco (4.2%)</i> <i>Fabricated metal</i> products (1.2%)

46

Trade integration and production sharing: a characterization...

Table 3	(continued)
---------	-------------

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Colombia	Chemical products (44.3%) ^b Rubber and plastics products (33.7%) Agriculture (4.3%) Other non-metallic mineral products (1.4%) ^b Mining and quarrying (0.0%)	Chemical products (44.8%) Rubber and plastics products (24.5%) Other non-metallic mineral products (3.9%) ^b Food products, beverages and tobacco (3.7%) Mining and quarrying (0.0%) ^b	Mining and quarrying (0.3%) ^b		Chemical products (25.4%) ^b Rubber and plastics products (21.4%) Other non-metallic mineral products (0.7%) ^b Mining and quarrying (0.4%) Wood and wood products (0.1%) ^b	Rubber and plastics products (26.9%) ^b Chemical products (20.0%) ^b Textiles (18.4%) Other non-metallic mineral products (3.6%) ^b Food products, beverages and tobacco (3.1%) Mining and quarrying (2.9%) ^b	Chemical products (38.6%) Textiles (19.0%) ^b Pulp, paper and paper products (13.1%) ^b Rubber and plastics products (10.4%) ^b Electrical machinery and apparatus, n.e.c. (2.3%) ^b Other non-metallic mineral products (0.3%) ^b Mining and quarrying (0.3%)
Ecuador	Food products, beverages and tobacco (69.4%)	Rubber and plastics products (23.4%) Wood and wood products (19.4%) ^b Food products, beverages and tobacco (10.4%) ^b <i>Textiles (8.1%)</i> Agriculture (6.8%) ^b	Food products, beverages and tobacco (18.9%) ^b Mining and quarrying (0.0%)	Food products, beverages and tobacco (21.3%) <i>Rubber and plastics</i> <i>products</i> (11.2%) Agriculture (9.4%) <i>Textiles</i> (8.7%) <i>Pulp, paper and</i> <i>paper products</i> (5.8%) ^b Wood and wood products (5.3%) Other non-metallic mineral products (0.7%)	Fabricated metal products (34.6%) ^b Chemical products (32.1%) ^b Food products, beverages and tobacco (3.7%) ^b Other non-metallic mineral products (1.5%)	Agriculture (41.5%) ^b Wood and wood products (12.8%) Food products, beverages and tobacco (9.2%)	Wood and wood products (8.0%) Mining and quarrying (3.2%)

Table 3	(continued)
---------	-------------

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Paraguay	Agriculture (24.8%) <i>Rubber and plastics</i> <i>products</i> (12.8%) ^b Food products, beverages and tobacco (11.1%) <i>Pulp, paper and</i> <i>paper products</i> (9.1%) ^b Textiles (8.4%) Wood and wood products (4.1%) Other non-metallic <i>mineral products</i> (2.7%)	Agriculture (59.2%) <i>Rubber and plastics</i> <i>products</i> (13.1%) ^b Food products, beverages and tobacco (11.5%) Textiles (2.1%) Wood and wood products (0.9%)	Food products, beverages and tobacco (50.0%) Agriculture (41.2%) Wood and wood products (2.5%)	Agriculture (94.1%) Food products, beverages and tobacco (4.0%)	Wood and wood products (1.3%) ^b	Agriculture (82.8%) Textiles (10.1%) Wood and wood products (4.5%) Food products, beverages and tobacco (1.4%)	Food products, beverages and tobacco (64.5%) Agriculture (35.3%) ^b
Peru	Mining and quarrying (33.9%) Basic metals (23.1%) <i>Rubber and plastics</i> <i>products</i> (6.8%) ^b Agriculture (4.7%) <i>Textiles</i> (4.6%) Food products, beverages and tobacco (1.8%) <i>Other non-metallic</i> <i>mineral products</i> (0.9%) ^b	Basic metals (72.2%) Mining and quarrying (20.6%) <i>Textiles (0.7%)</i>	Mining and quarrying (82.2%) Food products, beverages and tobacco (3.8%) Textiles (1.2%) Other non-metallic mineral products (0.3%)	Basic metals (54.0%) Rubber and plastics products (12.0%) ^b Mining and quarrying (6.4%) Textiles (3.8%) Food products, beverages and tobacco (3.4%) Pulp, paper and paper products (3.4%) ^b Agriculture (3.4%) ^b Other non-metallic mineral products (1.0%)	Rubber and plastics products (43.0%) ^b Electrical machinery and apparatus, n.e.c. (6.2%) ^b Agriculture (5.1%) ^b Food products, beverages and tobacco (3.6%) Other non-metallic mineral products (0.7%)	Basic metals (29.7%) Mining and quarrying (24.3%) Wood and wood products (15.1%) Textiles (3.9%) ^b Other non-metallic mineral products (1.3%) ^b	

Table 3	(continued)
---------	-------------

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Uruguay	Motor vehicles (22.5%) Rubber and plastics products (13.4%) Pulp, paper and paper products (9.5%) <i>Electrical</i> machinery and apparatus, n.e.c.(6.2%) ^b Food products, beverages and tobacco (4.9%) Textiles (2.8%) Wood and wood products (0.9%) ^b Other non-metallic mineral products (0.8%)	Rubber and plastics products (35.4%) Food products, beverages and tobacco (25.5%) Agriculture (19.4%) Pulp, paper and paper products (1.6%) ^b Wood and wood products (0.2%) ^b	Rubber and plastics products (25.4%) <i>Chemical products</i> (23.7%) ^b Wood and wood products (17.5%) ^b Pulp, paper and paper products (9.7%) ^b Textiles (7.8%)	Chemical products (42.3%) ^b Food products, beverages and tobacco (9.0%) Textiles (7.9%) Agriculture (7.6%) Wood and wood products (7.2%) ^b Machinery and equipment, n.e.c. (5.0%) ^b	Fabricated metal products (45.4%) ^b Chemical products (26.6%) Other transport equipment (1.6%) ^b	Chemical products (49.2%) ^b Textiles (20.5%) Wood and wood products (12.3%) ^b Food products, beverages and tobacco (1.0%)	Textiles (24.3%) Wood and wood products (5.9%) ^b
Venezuela (Bolivarian Republic of)	Fabricated metal products (47.2%) Chemical products (31.3%) Basic metals (8.1%) ^b Agriculture (1.2%) ^b	Basic metals (78.1%) ^b Other non-metallic mineral products (1.4%)	Chemical products (55.1%) Basic metals (38.4%) ^b Fabricated metal products (4.6%)	Basic metals (50.9%) Chemical products (33.8%) ^b Fabricated metal products (5.2%) Wood and wood products (0.8%) ^b Other non-metallic mineral products (0.4%)	Basic metals (72.9%) Chemical products (21.7%) ^b Other non-metallic mineral products (0.3%) ^b	Basic metals (71.4%) Chemical products (19.2%) Mining and quarrying (3.2%)	Basic metals (73.7%) Chemical products (19.9%) Fabricated metal products (2.2%) Mining and quarrying (0.0%)

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Mexico and C	Central America			•	•		
Mexico	Computer, electronic and optical equipment (9.0%) ^b	Basic metals (35.0%) Chemical products (26.4%) Motor vehicles (4.9%) Other non-metallic mineral products (2.7%)	Basic metals (20.0%) Chemical products (13.8%) Computer, electronic and optical equipment (1.8%) ^b Other non-metallic mineral products (1.2%) Electrical machinery and apparatus, n.e.c. (1.2%) ^b	Basic metals (41.2%) ^b Electrical machinery and apparatus, n.e.c. (7.8%) Other non-metallic mineral products (0.8%)	Basic metals (32.8%) Chemical products (17.0%) Fabricated metal products (15.6%) Rubber and plastics products (14.4%) Electrical machinery and apparatus, n.e.c. (5.6%) Textiles (2.0%) ^b Food products, beverages and tobacco (1.4%) Other non-metallic mineral products (1.1%)		Basic metals (47.8%) Computer, electronic and optical equipment (2.2%) ^b Electrical machinery and apparatus, n.e.c. (1.5%) ^b Motor vehicles (0.9%) ^b Other non-metallic mineral products (0.4%)
Costa Rica	Computer, electronic and optical equipment (59.2%) ^b Rubber and plastics products (14.6%) ^b <i>Textiles (4.3%)^b</i>	Computer, electronic and optical equipment (84.0%) ^b Rubber and plastics products (6.5%) ^b Agriculture (4.7%) ^b Wood and wood products (0.4%) ^b	Basic metals (8.2%) Rubber and plastics products (6.0%) ^b Computer, electronic and optical equipment (2.0%) ^b Agriculture (1.8%) Other non-metallic mineral products (0.8%) ^b Textiles (0.8%) ^b Electrical machinery and apparatus, n.e.c. (0.3%) ^b	Electrical machinery and apparatus, n.e.c. (15.5%) ^b Rubber and plastics products (13.2%) Food products, beverages and tobacco (9.4%) ^b Computer, electronic and optical equipment (8.3%) ^b <i>Mining and</i> <i>quarrying (4.5%)</i> <i>Textiles (2.5%)</i>		Computer, electronic and optical equipment (92.1%) ^b Food products, beverages and tobacco (1.3%)	Computer, electronic and optical equipment (41.2%) ^b Rubber and plastics products (22.1%) ^b Electrical machinery and apparatus, n.e.c. (5.3%) Fabricated metal products (2.5%) ^b Other non-metallic mineral products (0.6%) ^b

Table 3	(continued)
Tuble 0	(continueu)

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
El Salvador	Rubber and plastics products (65.6%) ^b Pulp, paper and paper products (27.5%) Textiles (4.0%) ^b Agriculture (2.0%) ^b	Textiles (6.9%) ^b Rubber and plastics products (15.9%) ^b		Basic metals (58.9%) Textiles (8.2%)	Basic metals (36.6%) ^b Rubber and plastics products (21.4%) ^b Pulp, paper and paper products (13.4%) ^b Chemical products (10.5%) ^b Textiles (4.1%)	Textiles (39.9%) Pulp, paper and paper products (21.4%) Rubber and plastics products (14.8%)	Basic metals (64.5%) ^b Pulp, paper and paper products (9.0%) ^b Fabricated metal products (2.3%) ^b Computer, electronic and optical equipment (0.8%) Other non-metallic mineral products (0.5%) ^b
Guatemala	Agriculture (68.7%) Food products, beverages and tobacco (29.5%) ^b	Basic metals (42.4%) Agriculture (36.0%)	Food products, beverages and tobacco (32.9%) Agriculture (13.5%) ^b	Agriculture (54.1%) Food products, beverages and tobacco (21.7%) ^b	Basic metals (32.5%) ^b Rubber and plastics products (27.6%) Fabricated metal products (13.1%) ^b Agriculture (6.7%) ^b Pulp, paper and paper products (5.1%) ^b Textiles (2.1%) Other non-metallic mineral products (1.0%) Food products, beverages and tobacco (0.9%) Wood and wood products (0.1%)	Textiles (33.8%) Food products, beverages and tobacco (23.8%) Agriculture (15.7%) <i>Wood and wood</i> <i>products (2.6%)</i>	Agriculture (54.7%) ^b Food products, beverages and tobacco (35.8%)

Table 3	(continued)
---------	-------------

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Honduras	Food products, beverages and tobacco (21.7%) ^b	Fabricated metal products (11.6%) ^b Electrical machinery and apparatus, n.e.c. (8.9%) ^b Agriculture (6.5%) ^b Pulp, paper and paper products (2.7%) ^b Wood and wood products (0.7%) ^b	Fabricated metal products (0.4%) ^b Chemical products (0.4%) Food products, beverages and tobacco (0.4%) ^b Textiles (0.3%) Wood and wood products (0.1%) ^b	Food products, beverages and tobacco (65.9%) ^b	Pulp, paper and paper products (27.6%) ^b Rubber and plastics products (26.3%) ^b Fabricated metal products (21.0%) Textiles (5.4%) Agriculture (4.7%) Food products, beverages and tobacco (1.7%) ^b Wood and wood products (0.2%)	Electrical machinery and apparatus, n.e.c. (44.4%) ^b <i>Mining and quarrying</i> (21.1%) Food products, beverages and tobacco (14.7%)	Food products, beverages and tobacco (44.0%) Textiles (18.1%) ^b Other non- metallic mineral products (1.7%) ^b Other transport equipment (1.1%) ^b
Nicaragua		Agriculture (43.9%) ^b Food products, beverages and tobacco (39.8%) ^b	Other non-metallic mineral products (0.5%) ^b	Agriculture (89.2%) ^b	Rubber and plastics products (33.8%) Food products, beverages and tobacco (19.6%) Agriculture (9.5%) Wood and wood products (0.1%)	Electrical machinery and apparatus, n.e.c. (79.8%) ^b Agriculture (12.7%) Food products, beverages and tobacco (2.8%)	Computer, electronic and optical equipment (29.5%) ^b Other non- metallic mineral products (1.8%)
Panama	Chemical products (62.1%)	Chemical products (47.7%) Textiles (7.1%)	Motor vehicles (2.5%) ^b Textiles (1.6%) Computer, electronic and optical equipment (1.0%) ^b	Textiles (26.8%) Computer, electronic and optical equipment (11.8%)	Rubber and plastics products (28.1%) ^b Textiles (4.4%) ^b Food products, beverages and tobacco (1.7%) Wood and wood products (0.2%) ^b	Chemical products (49.4%) Textiles (14.2%) Other non-metallic mineral products (6.9%) ^b	Computer, electronic and optical equipment (9.2%) ^b
The Caribbean	1		1	1			
Antigua and Barbuda		Fabricated metal products (80.7%) ^b Other non-metallic mineral products (19.3%) ^b		Machinery and equipment, n.e.c. (24.9%) ^b	Chemical products (56.5%) Fabricated metal products (40.8%) ^b	Computer, electronic and optical equipment (72.6%) ^b Machinery and equipment, n.e.c.(12.3%) ^b Electrical machinery and apparatus, n.e.c. (8.3%) ^b	

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Bahamas	Chemical products (99.3%)		Fabricated metal products (31.8%) ^b		Rubber and plastics products (53.7%) ^b		
Barbados	Machinery and equipment, n.e.c. (68.9%) <i>Motor vehicles</i> (31.1%) ^b	Mining and quarrying (47.9%) ^b		Chemical products (81.7%) ^b Mining and quarrying (12.0%) ^b	Pulp, paper and paper products (70.5%) Other non-metallic mineral products (22.5%) ^b	Electrical machinery and apparatus, n.e.c. (52.1%) Computer, electronic and optical equipment (37.8%) ^b	Chemical products (58.7%) ^b Basic metals (39.2%) ^b Wood and wood products (0.6%) ^b
Belize		Machinery and equipment, n.e.c. (89.7%) ^b Electrical machinery and apparatus, n.e.c. (4.2%) ^b Other non-metallic mineral products (0.7%) ^b		Rubber and plastics products (100%) ^b	Mining and quarrying (0.1%) ^b	Wood and wood products (4.0%)	Fabricated metal products (100%) ^b
Cuba	Chemical products (99.2%)	Chemical products (88.9%) Other non-metallic mineral products (10.6%)		Basic metals (49.4%) <i>Chemical products</i> (38.8%) Food products, beverages and tobacco (8.6%) ^b <i>Electrical machinery</i> <i>and apparatus, n.e.c.</i> (2.8%)	Fabricated metal products (68.9%) ^b Machinery and equipment, n.e.c. (9.8%) ^b Food products, beverages and tobacco (8.7%) ^b Agriculture (6.6%) ^b	Basic metals (47.6%) Chemical products (14.9%) Food products, beverages and tobacco (3.0%)	Chemical products (97.8%)
Dominica	Chemical products (100%) ^b	Fabricated metal products (73.2%) ^b Machinery and equipment, n.e.c. (18.0%) ^b Chemical products (8.8%) ^b	Electrical machinery and apparatus, n.e.c. (100%) ^b	Mining and quarrying (96.1%) ^b Electrical machinery and apparatus, n.e.c. $(1.7\%)^b$ Machinery and equipment, n.e.c. $(1.0\%)^b$ Pulp, paper and paper products $(0.9\%)^b$	Rubber and plastics products (62.0%) ^b Pulp, paper and paper products (9.3%) ^b Machinery and equipment, n.e.c. (7.7%) ^b	Electrical machinery and apparatus, n.e.c. (37.0%) ^b Computer, electronic and optical equipment (26.7%) ^b Fabricated metal products (20.6%) ^b Rubber and plastics products (7.9%) ^b Textiles (4.6%) ^b	

Table 3 (continued)

Trade integration and production sharing: a characterization...

ECLAC - International Trade Series No. 137

Table 3 (continued)
-----------	------------

54

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Dominican Republic	Electrical machinery and apparatus, n.e.c. (24.8%)	Electrical machinery and apparatus, n.e.c. (27.9%) ^b Other transport equipment (26.3%) ^b Rubber and plastics products (7.2%)	Electrical machinery and apparatus, n.e.c. (0.6%) ^b	Mining and quarrying (15.4%) ^b Other non-metallic mineral products (1.9%) Wood and wood products (1.1%) ^b	Basic metals (57.2%) ^b Fabricated metal products (18.6%) ^b Rubber and plastics products (10.4%) Electrical machinery and apparatus, n.e.c. (3.0%) ^b Food products, beverages and tobacco (1.5%) ^b Other non-metallic mineral products (0.2%) ^b	Electrical machinery and apparatus, n.e.c. (47.1%) ^b Textiles (18.3%) ^b Rubber and plastics products (16.7%) ^b Wood and wood products (0.9%) ^b	Other non-metallic mineral products (48.5%) ^b Electrical machinery and apparatus, n.e.c. (23.3%) ^b
Grenada		Electrical machinery and apparatus, n.e.c. (79.9%) ^b Motor vehicles (20.1%) ^b			Computer, electronic and optical equipment (91.4%) ^b Agriculture (6.3%) ^b Electrical machinery and apparatus, n.e.c. (2.3%) ^b	Electrical machinery and apparatus, n.e.c. (47.1%) ^b Basic metals (27.6%) ^b Fabricated metal products (20.3%) ^b	Fabricated metal products (100%) ^b
Guyana		Food products, beverages and tobacco (81.8%) Machinery and equipment, n.e.c. (8.8%) ^b	Mining and quarrying (100%)	Mining and quarrying (99.9%) Wood and wood products (0.1%)	Mining and quarrying (87.6%) ^b Wood and wood products (12.4%)	Mining and quarrying (88.0%) ^b Agriculture (1.7%) ^b	Mining and quarrying (100%)
Haiti	Chemical products (98.8%) Other non-metallic mineral products (1.2%) ^b	Chemical products (65.4%) ^b		Chemical products (83.2%) ^b Computer, electronic and optical equipment (16.8%) ^b	Fabricated metal products (94.3%) ^b Textiles (2.8%) ^b Agriculture (2.7%) ^b		Pulp, paper and paper products (100%)

Table 3	(continued)
---------	-------------

Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Jamaica	Chemical products (72.5%) ^b Agriculture (27.5%) ^b	Food products, beverages and tobacco (94.0%) ^b	Wood and wood products (55.5%) ^b	Mining and quarrying (78.9%) Chemical products (14.3%) ^b Other non-metallic mineral products (1.1%) ^b	Other non-metallic mineral products (40.9%) ^b	Fabricated metal products (65.9%) ^b	
Saint Kitts and Nevis	Electrical machinery and apparatus, n.e.c. (100%) ^b	Electrical machinery and apparatus, n.e.c. (99.8%) ^b				Computer, electronic and optical equipment (92.3%) ^b Electrical machinery and apparatus, n.e.c. (7.7%)	Electrical machinery and apparatus, n.e.c. (100%) ^b
Saint Lucia	Computer, electronic and optical equipment (100%) ^b	Computer, electronic and optical equipment (91.3%) ^b			Other non-metallic mineral products (75.2%) ^b Textiles (24.8%) ^b	Computer, electronic and optical equipment (62.4%) ^b <i>Electrical</i> <i>machinery and</i> <i>apparatus, n.e.c.</i> (37.3%) ^b	Computer, electronic and optical equipment (100%) ^b
Saint Vincent and the Grenadines	Computer, electronic and optical equipment (100%) ^b						
Suriname		Machinery and equipment, n.e.c. (43.7%) ^b Electrical machinery and apparatus, n.e.c. (23.5%) ^b		Fabricated metal products (74.2%) ^b Computer, electronic and optical equipment (25.6%) ^b	Other non-metallic mineral products (19.6%) ^b Pulp, paper and paper products (15.6%) ^b	Fabricated metal products (27.4%) ^b	Mining and quarrying (85.6%) ^ь

ECLAC - International Trade Series No. 137

Table 3 (conclude	ed)						
Exporting country	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Trinidad and Tobago	Mining and quarrying (66.1%) ^b	Chemical products (49.9%) Basic metals (46.7%) ^b Mining and quarrying (2.2%) ^b	Chemical products (97.3%) Mining and quarrying (0.0%) ^b	Chemical products (71.4%) Mining and quarrying (8.8%) ^b	Basic metals (97.0%) Mining and quarrying (0.0%) ^b	Chemical products (66.4%) ^b Basic metals (26.2%) Mining and quarrying (6.6%) ^b	Basic metals (97.8%)

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

^a Industries are ranked, from highest to lowest, on the basis of their share in countries' vertical specialization-related intermediate exports to the destination market (indicated in brackets). In italics, industries in which the exporting country has a RCA in the destination market but not in the world market.

^b Industries that were not competitive in 1995-1996.

IV. Concluding remarks

Latin American and Caribbean countries have been poorly integrated into international production networks. Particularly, the analysis carried out in this document shows that in 1995-2011 the region's participation as a source of foreign intermediate goods for the three major GVCs (Factory North America, Factory Europe, and Factory Asia) was limited and concentrated in a few origin countries (in decreasing order of importance, Brazil, Mexico, Chile, Argentina, Peru, and Costa Rica). In addition, the level of intra-regional production integration is considerably lower than that of other regions. However, both current and potential engagement in intra and extra-regional production networks vary greatly across countries.

The main findings on Latin America and the Caribbean's gross forward integration into extraregional value chains are:

- The concentration of gross forward linkages increased since 1995 in terms of destination markets, due to the higher participation of Factory Asia (particularly, China) as destination of Latin American and Caribbean countries' exports. Within the region, Mexico —highly dependent on the United States market— and, to a lesser extent, Central America and the Caribbean are less diversified than the South American subregion. Along with Mexico, Chile and Costa Rica have higher levels of market concentration than the other three main origin countries (Argentina, Brazil, and Peru).
- On the sector dimension, the concentration of gross forward linkages is considerably higher for Central America and the Caribbean than for Mexico and South America. Also, exports to Factory Asia are generally more concentrated than those oriented to Factory North America and Factory Europe (the main exception is the South American subregion). Among the six main origin countries, Costa Rica and Chile show higher and increasing concentration levels, although in different sectors (natural resource-based products for Chile and high-technology products for Costa Rica).
- Gross forward linkages with Factory Europe and Factory Asia concentrate in primary sectors (Mining and quarrying and Agriculture). Links with Factory North America are more diversified in terms of origin sectors (with Basic metals, Mining and quarrying, Motor vehicles and Electrical machinery and apparatus, n.e.c. as the main source industries).
- The trade agreements signed with GVC members would not have contributed much to promote Latin American and Caribbean countries' insertion into GVCs (although causality is not evaluated). Four of the six main origin countries (Chile, Costa Rica, Mexico, and Peru) gained preferential access to at least one of the three GVCs in 1995-2011, with mixed results in terms

of export diversification. For other countries with preferential access to those markets, the share in the region's gross forward linkages remained very low (or even decreased).

• In terms of export potential, Latin American and Caribbean countries' revealed competitiveness as a source of foreign inputs for extra-regional value chains concentrate in primary sectors and low and medium-low-technology industries (most of which do not have a significant participation in the imported inputs embodied in the GVCs' exports). Although some countries (like Brazil, Costa Rica, the Dominican Republic, Honduras, Nicaragua, and Mexico) show competitiveness in some of the most technology-intensive industries (Computer, electronic and optical equipment; Electrical machinery and apparatus, n.e.c.; or Motor vehicles), these industries account for a very low share of the region's forward linkages with the GVCs.

The analysis of the import content of Argentina, Brazil, Chile, Colombia, Costa Rica, Mexico and Peru's manufacturing exports shows that:

- Brazil is characterized by a relatively low level of gross vertical specialization, in comparison with other Latin American and extra-regional countries. The other four natural resource-intensive economies (Argentina, Chile, Colombia, and Peru) have similar (intermediate) levels of gross backward linkages. Costa Rica and Mexico, who specialize in assembling and processing foreign inputs, show the highest levels. In all seven countries, aggregate vertical specialization levels hide large differences across exporting industries.
- Brazil and Costa Rica have lower levels of intra-regional backward linkages than the other countries (except for Mexico, which is integrated into the North America region). Although most imported inputs in Argentina, Chile, Colombia and Peru's manufacturing exports come from outside the region (particularly, the United States, the European Union, China and the Rest of South and East Asia), Latin American and Caribbean countries have a significant participation. Argentina stands out for its high level of backward integration with Brazil. Other important intra-regional linkages include those between Argentina and Brazil —the largest MERCOSUR members— and the Pacific Alliance countries (particularly, Chile, Mexico and Peru). Also, significant intra-Pacific Alliance linkages are observed.
- Due to the competition of China, the United States and the European Union lost participation in Latin America's gross backward linkages. This is particularly the case of the technology-intensive sectors Computer, electronic and optical equipment and Electrical machinery and apparatus, n.e.c., where the participation of China reached high levels. However, in these industries a large proportion of the value added embodied in Chinese gross exports is sourced from abroad (mostly from other Southeast Asian countries, the European Union and the United States).
- Latin American and Caribbean countries have a larger participation as intra-regional providers
 of low and medium-low-technology inputs (including Food products, beverages and tobacco,
 Pulp, paper and paper products, Rubber and plastics products, and Basic metals). Their share is
 also important in the primary sectors (Agriculture and Mining and quarrying). However, a few
 countries have significant participations in the most technology-intensive industries (like Brazil
 in Motor vehicles, Electrical machinery and apparatus, n.e.c. and Computer, electronic and
 optical equipment for Argentina; Argentina in Motor vehicles for Brazil; and Brazil and Mexico
 in Electrical machinery and apparatus, n.e.c. and Motor vehicles for Costa Rica).
- Intra-regional competitiveness in intermediate goods concentrates in primary sectors and low and medium-low-technology industries. Countries like Argentina, Brazil, Costa Rica, the Dominican Republic, Honduras, Mexico, Nicaragua, Panama and Uruguay also show competitiveness, in some regional markets, in medium-high or high-technology industries (Machinery and equipment, n.e.c., Electrical machinery and apparatus, n.e.c., Computer, electronic and optical equipment, Motor vehicles or Other transport equipment, depending on the country). However, the share of these industries in the region's gross forward linkages with Argentina, Brazil, Chile, Colombia, Costa Rica, Mexico and Peru is low.

• For many countries, comparative advantages in some industries —particularly, those with higher technology content— are limited to the regional level. This would be related to countries' preferential access and geographical proximity to regional markets, which shows the importance of regional integration.

The comparison of intra and extra-regional gross forward linkages reveals that:

For South American countries, intra-regional linkages are somewhat less concentrated in primary and low and medium-low-technology inputs than their linkages with Factory North America, Factory Europe and, particularly, Factory Asia (see figure 23).⁴⁵ Also, they are overall more diversified in the sector dimension (see figure 24). However, the share of Latin America (6 countries) and Mexico as destination of South American vertical specialization-related intermediate exports is, at the aggregate level, quite low (9% and 3% respectively, in 2011). Also, both intra and extra-regional linkages are concentrated in a few origin countries (Brazil, Chile, Argentina and Peru, with around 90% of the total).





Agriculture Mining Low & medium-low-technology industries Medium-high & high-technology industries

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

^a Latin America (6 countries) includes Argentina, Brazil, Chile, Colombia, Costa Rica and Peru. Mexico is included in Factory North America. Percentages in brackets next to the name of each exporting subregion/country indicate the share of the corresponding destination (i.e., Latin America (6 countries), Factory North America, Factory Europe, or Factory Asia) in the subregion/country's total vertical specialization-related intermediate exports in 2011.

• Unlike South America, low and medium-low-technology industries account for a larger proportion of Mexico's forward linkages with Latin America (6 countries) than of its extraregional linkages. In addition, intra-regional linkages (only 3% of the total in 2011) show a somewhat higher overall level of sector concentration. Similarly, for Central American countries, the most technology-intensive industries have a significantly larger participation in extra-regional linkages (which is mostly explained by Costa Rica's forward integration into

⁴⁵ The share of medium-high or high-technology industries is higher in South American countries' gross forward linkages with Argentina, Colombia and Mexico than in those with Brazil, Chile, Costa Rica and Peru.

Factory North America and Factory Asia).⁴⁶ Central America's linkages with Latin America (6 countries) are, however, considerably more diversified than its extra-regional linkages in terms of both origin sectors and countries.

• The Caribbean countries' forward linkages with Latin America (6 countries) (3% of the total in 2011) are similar in composition to those with Factory North America, and are less concentrated in low and medium-low-technology industries than their linkages with Factory Europe and, particularly, Factory Asia. However, these intra-regional linkages show a higher overall level of sector concentration than extra-regional linkages (except for Factory Asia), and they are highly concentrated in terms of origin countries (with Trinidad and Tobago and the Dominican Republic accounting for over 80% of the total).

Figure 24 Latin America and the Caribbean (selected subregions and countries): Herfindahl-Hirschman index of sector concentration of vertical specialization-related intermediate exports, 2010-2011^a



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Latin America (6 countries) includes Argentina, Brazil, Chile, Colombia, Costa Rica and Peru. Mexico is included in Factory North America.

Although the limited forward engagement of most Latin American and Caribbean countries in extra-regional value chains could be related to structural factors such as countries' distance to the main manufacturing hubs in Factory North America, Factory Europe and Factory Asia, the data presented in this document show the potential for further integration. Also, the relatively low level of intra-regional links suggests the scope for further development of regional production networks, exploiting production complementarities within the region.

Integration into international value chains —and the division of production and task specialization involved— brings benefits beyond those traditionally associated with trade in final goods (Cheng et al., 2015). It offers opportunities for deeper specialization and greater export and production diversification, enabling countries to exploit finer comparative advantage niches and benefit from economies of scale and scope. Through economic upgrading and the engagement of more local actors (firms and workers), participation in international value chains can be a path for faster industrialization and development (Taglioni and Winkler, 2016). However, an integration strategy that maximizes the positive spillovers into the domestic economy —in terms of productivity gains, creation of high-quality jobs, skills and technology transfer to local firms, and greater integration of small and medium-sized enterprises— requires coordinated plans across a number of policy areas (like trade, FDI, innovation and human capital formation).

⁴⁶ It should be noted that medium-high or high-technology industries have a considerably larger participation in Mexico and Central America's forward linkages with Argentina and Brazil than in their linkages with the rest of Latin America (6 countries).

As shown by Cadestin et al. (2016), trade barriers on intermediate imports affect countries' engagement in cross-border production sharing. The multiple intra and extra-regional trade agreements implemented by Latin American and Caribbean countries have significantly lowered tariffs.⁴⁷ However, the overlapping and competing rules of origin governing trade under those agreements —by establishing when a product is eligible for preferential access— impose limits on the range of countries that can participate in the international fragmentation of production (affecting both intra and extra-agreement input suppliers). The harmonization of existing rules of origin and amendments to certification, *de minimis* and cumulation rules would contribute to stimulate both intra and extra-agreement value chains. Also, the trade-distorting effect of some standards, technical regulations, conformity-assessment procedures and similar non-tariff measures (NTMs) need to be addressed —through mutual recognition or harmonization—, to improve Latin American and Caribbean countries' integration into intra and extra-regional value chains.

A structural factor like geography imposes, however, a challenge for Latin American and Caribbean countries to join value chains in other regions, or even to develop value chains within their own region (Blyde, 2014). The low quality of logistics infrastructure (transport, storage and distribution) and other services required to move intermediate goods or coordinate production across countries (like finance and ICT services) affects Latin American and Caribbean countries' participation in international value chains. Addressing long-standing shortcomings in the region's logistics infrastructure, to improve international connectivity and lower coordination and transport costs, is therefore crucial for being able to better exploit the oportunities offered by a globalized world economy.⁴⁸

Other factors affecting countries' integration into international production networks are the legal and institutional framework for FDI, intellectual property rights and contract enforcement, and the efficiency of customs procedures (Blyde, 2014). International fragmentation of production involves cross-border investments (particularly, vertical FDI that links affiliates providing inputs to parent companies in other countries), which might be inhibited by inadequate investment rules in the host country. Likewise, the delegation of links of the value chain to foreign suppliers through contract manufacturing —and the related knowledge and technology transfer— can be undermined by uncertain and ambiguous practices in contract enforcement, as well as by weak intellectual property rights. In addition, the administrative trade barriers imposed by cumbersome customs procedures may inhibit the rapid delivery of components, hampering countries' engagement in cross-border production sharing.

Along with increasing their participation in international value chains, Latin American and Caribbean countries need to upgrade within them in order to maximize the positive spillovers for local actors. This implies gaining competitiveness in more technology-intensive and higher value-added products (product upgrading), tasks (functional upgrading) and/or sectors (inter-sector upgrading), which could be achieved by improving the skills and know-how of the workforce (skills upgrading), improving firms' absorptive capacity and technology (capital upgrading), or increasing productivity in existing tasks (process upgrading) (Taglioni and Winkler, 2016). For many countries, upgrading should also address the issue of environmental sustainability, by shifting the composition of their forward linkages towards cleaner industries and tasks.

Industrial policy plays a key role in improving countries' ability to enhance their upgrading opportunities within international value chains. As pointed out by Gereffi and Sturgeon (2013), the central goal of a GVC-oriented industrial policy is not creating fully vertically integrated national industries but moving into higher-value niches in GVCs. Such policy involves actions in areas like technological progress and innovation, human capital formation, competitiveness, and competition.

⁴⁷ Since tariff costs are incurred repeatedly as goods in process cross multiple borders, they have a proportionally larger effect on trade involving goods produced sequentially in multiple countries than on trade in goods produced entirely in one country (Hummels et al., 2001). Thus, even small further reductions in tariff barriers would promote Latin American and Caribbean countries' vertical specialization-related trade.

⁴⁸ Latin America and the Caribbean has, on average, less adequate logistics infrastructure than other regions. Moreover, logistics deficiencies make freight costs for intra-regional exports almost as expensive as they are for extra-regional exports, or even more. This affects the region's ability to attract fragmented production, particularly in industries that are more dependent on logistics services (CAF/ECLAC/OECD, 2013; Blyde, 2014).

As seen above, a comprehensive policy aimed at promoting countries' participation in intra and extraregional value chains would involve measures that can only be fully implemented over long periods (like the improvement of transport infrastructure or the development of higher levels of human capital); however, other measures can be undertaken in a shorter time horizon (like trade facilitation measures). Likewise, some measures can be implemented unilaterally while others require negotiation and cooperation between countries (for instance, through deeper integration). Overall, policy measures have an important role to play in order to compensate Latin American and Caribbean countries' relative disadvantage in structural factors such as distance to the major GVC regions and across the continent.

References

- Ahmad, N., Bohn, T., Mulder, N., Vaillant, M. and Zaclicever, D. (2017). "Indicators on global value chains: A guide for empirical work", OECD Statistics Working Papers 2017/08, OECD Publishing, Paris.
- Balassa, B. (1965). "Trade liberalization and 'revealed' comparative advantage", *The Manchester School* 33 (2): 99-123.
- Baldwin, R. (2006). "Globalisation: The Great Unbundling(s)", Economic Council of Finland 20: 5-47.
- (2013). "Trade and industrialization after globalization's second unbundling: How building and joining a supply chain are different and why it matters", in Feenstra, R.C. and Taylor, A.M. (eds.), *Globalization in* an age of crisis: Multilateral economic cooperation in the twenty-first century, University of Chicago Press.
- Baldwin, R. and Lopez-Gonzalez, J. (2015). "Supply-chain trade: A portrait of global patterns and several testable hypotheses", *The World Economy* 38 (11): 1682-1721.
- Blyde, J.S. (ed.) (2014). Synchronized Factories. Latin America and the Caribbean in the Era of Global Value Chains, Springer International Publishing.
- Cadestin, C., Gourdon, J. and Kowalski, P. (2016), "Participation in global value chains in Latin America: Implications for trade and trade-related policy", *OECD Trade Policy Papers* No. 192, OECD Publishing, Paris.
- CAF/ECLAC/OECD (Development Bank of Latin America/Economic Commission for Latin America and the Caribbean/Organization for Economic Cooperation and Development) (2013). Latin American Economic Outlook 2014. Logistics and Competitiveness for Development (LC/G.2575), Paris.
- Cheng, K., Rehman, S., Seneviratne, D. and Zhang, S. (2015). "Reaping the benefits from global value chains", *IMF Working Paper* WP/15/204, International Monetary Fund.
- De Backer, K. and Yamano, N. (2012). "International Comparative Evidence on Global Value Chains", *OECD Science, Technology and Industry Working Papers* 2012/03, OECD Publishing.
- Dür, A., Baccini, L. and Elsig, M. (2014). "The Design of International Trade Agreements: Introducing a New Database", *Review of International Organizations* 9 (3): 353-375.
- ECLAC (Economic Commission for Latin America and the Caribbean) (2016). *Latin America and the Caribbean in the World Economy 2016: The region amid the tensions of globalization* (LC/G.2697-P), Santiago.
- Estevadeordal, A., Blyde, J. and Suominen, K. (2013). *Are global value chains really global? Policies to accelerate countries' access to international production networks*, E15Initiative. Geneva: International Centre for Trade and Sustainable Development (ICTSD) and World Economic Forum.
- Gamberoni, E., Lanz, R. and Piermartini, R. (2010). "Timeliness and contract enforceability in intermediate goods trade", *World Bank Policy Research Working Paper* No. 5482.

- Gaulier, G. and Zignago, S. (2010). "BACI: International trade database at the product-level The 1994-2007 version", *CEPII Working Paper* 2010-23.
- Gereffi, G. and Sturgeon, T. (2013). "Global value chain-oriented industrial policy: the role of emerging economies", in Elms, D.K. and Low, P. (eds.), *Global value chains in a changing world*, WTO Publications.
- Hummels, D., Ishii, J. and Yi, K. (2001). "The nature and growth of vertical specialization in world trade", *Journal of International Economics* 54:75-96.
- Johnson, R. and Noguera, G. (2012). "Proximity and production fragmentation", *American Economic Review:* Papers & Proceedings 102 (3): 407-411.
- Kowalski, P., Lopez-Gonzalez, J., Ragoussis, A. and Ugarte, C. (2015). "Participation of Developing Countries in Global Value Chains: Implications for Trade and Trade-Related Policies", OECD Trade Policy Papers No. 179, OECD Publishing, Paris.
- Lopez-Gonzalez, J. (2012). "The Impact of free trade agreements on vertical specialisation", NCCR-Trade Working Paper No. 2012/36.
- Lopez-Gonzalez, J. and Holmes, P. (2011). "The nature and evolution of vertical specialisation: What is the role of preferential trade agreements?", *NCCR-Trade Working Paper* No. 2011/41.
- Miroudot, S., Lanz, R. and Ragoussis, A. (2009). "Trade in intermediate goods and services", *OECD Trade Policy Papers* No. 93, OECD Publishing.
- Orefice, G. and Rocha, N. (2014). "Deep integration and production networks: An empirical analysis", *The World Economy* 37 (1): 106-136.
- Organisation for Economic Co-operation and Development (OECD) (2015). "Developing countries participation in global value chains and its implications for trade and trade related policies", *OECD Trade Policy Paper* No. 179, OECD Publishing.
 - (2016). "Cardiac arrest or dizzy spell: why is world trade so weak and what policy do about it?", OECD Economic Policy Paper No. 18, OECD Publishing.
- Taglioni, D. and Winkler, D. (2016). *Making Global Value Chains Work for Development*, Trade and Development series, Washington, DC: World Bank.
- Timmer, M., Los, B., Stehrer, R. and de Vries, G. (2016). "An anatomy of the global trade slowdown based on the WIOD 2016 release", *GGDC Research Memorandum* 162.

Annex

Additional tables and figures

Group	Code	Description
Primary	C01T05	Agriculture, hunting, forestry and fishing
	C10T14	Mining and quarrying
Manufacturing	C15T16	Food products, beverages and tobacco
-	C17T19	Textiles, textile products, leather and footwear
	C20	Wood and products of wood and cork
	C21T22	Pulp, paper, paper products, printing and publishing
	C23	Coke, refined petroleum products and nuclear fuel
	C24	Chemicals and chemical products
	C25	Rubber and plastics products
	C26	Other non-metallic mineral products
	C27	Basic metals
	C28	Fabricated metal products
	C29	Machinery and equipment, n.e.c.
	C30.32.33	Computer, Electronic and optical equipment
	C31	Electrical machinery and apparatus, n.e.c.
	C34	Motor vehicles, trailers and semi-trailers
	C35	Other transport equipment
	C36T37	Manufacturing n.e.c.; recycling
Services	C40T41	Electricity, gas and water supply
	C45	Construction
	C50T52	Wholesale and retail trade; repairs
	C55	Hotels and restaurants
	C60T63	Transport and storage
	C64	Post and telecommunications
	C65T67	Financial intermediation
	C70	Real estate activities
	C71	Renting of machinery and equipment
	C72	Computer and related activities
	C73T74	R&D and other business activities
	C75	Public admin. and defence; compulsory social security
	C80	Education
	C85	Health and social work
	C90T93	Other community, social and personal services
	C95	Private households with employed persons

Table A.1 Industry breakdown of OECD's ICIO tables

	List of countries covered by the OECD's ICIO tables
Region	Countries
North America	Canada, United States
Latin America	Argentina, Brazil, Chile, Colombia, Costa Rica, Mexico, Peru
European Union	Austria, Belgium, Bulgaria, Croatia, Cyprus, Czech Republic, Denmark, Estonia Finland, France, Germany, Greece, Hungary, Ireland, Italy, Latvia, Lithuania Luxembourg, Malta, Netherlands, Poland, Portugal, Romania, Slovak Republic Slovenia, Spain, Sweden, United Kingdom
Rest of Europe	Iceland, Norway, Russian Federation, Switzerland
South East Asia	Brunei Darussalam, Cambodia, China, Hong Kong (Special Administrative Region of China), India, Indonesia, Japan, Malaysia, Philippines, Republic of Korea, Singapore, Thailand, Taiwan (Province of China), Viet Nam
Rest of Asia	Israel, Saudi Arabia, Turkey
Oceania	Australia, New Zealand
Africa	Morocco, South Africa, Tunisia

Table A.2 List of countries





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

Country	Latin America and the Caribbean	Rest of the World
Argentina	Uruguay (1974 [°] , 1991)	India (2009°)
	Ecuador (1984°, 2001°, 2005°)	Israel (2009 ⁵)
	P_{razil} (1987 ^b 1991)	
	$DiaZii (1907^2, 1991)$ Colombia (1988 ^b 2001 ^b 2005 ^b)	
	P_{eff} (1988 ^b 2001 ^b 2006 ^b)	
	Venezuela (Bolivarian Republic of) (1988 ^b	
	2001^{b} 2005 ^b 2012)	
	Paraguay (1989 ^b 1991)	
	Chile $(1991^{b}, 1996^{b})$	
	Bolivia (Plurinational State of) (1997 ^b)	
	Cuba (2000 ^b , 2007 ^b)	
Brazil	Uruguay (1976 ^b , 1991)	India (2009 ^b)
	Argentina (1987 ^b , 1991)	Israel (2009 ^b)
	Paraguay (1991)	
	Chile (1996 ^b)	
	Bolivia (Plurinational State of) (1997 ^b)	
	Colombia (1999 ^b , 2005 ^b)	
	Ecuador (1999 [°] , 2005 [°])	
	Peru (1999 ⁶ , 2006 ⁶)	
	Cuba (2000 ⁵ , 2007 ⁵)	
	Mexico (2003°)	
	$1999^{-}, 2005^{-}, 2012)$	
	Surinamo (2006 ^b)	
Chilo	Sumane (2000)	Canada (1997)
Chile	$\Delta ranting (1905, 1990)$	European Union (2003)
	Bolivia (Plurinational State of) (1993 ^b)	United Sates (2004)
	Colombia (1993 ^b 2009 2012)	Iceland (2004)
	Venezuela (Bolivarian Republic of) (1993 ^b)	Liechtenstein (2004)
	Ecuador (1994 ^b , 2010 ^b)	Norway (2004)
	Brazil (1996 ^b)	Switzerland (2004)
	Peru (1998 ^b , 2009, 2012)	Republic of Korea (2004)
	Mexico (1999, 2012)	New Zealand (2006)
	Costa Rica (2002)	China (2006)
	El Salvador (2002)	Brunei Darussalam (2006)
	Guatemala (2002)	Singapore (2006)
	Honduras (2002)	Japan (2007)
	Cuba (2008 ^b)	India (2007 ^b)
	Panama (2008)	Australia (2009)
	Nicaragua (2012)	Turkey (2011)
		Malaysia (2012)
		Hong Kong (2014)
<u></u>		Viet Nam (2014)
Colombia	Costa Rica (1985 ⁶ , 2016)	Canada (2011)
	El Salvador (1985 ⁶ , 2009)	United States (2012)
	Honduras (1985°, 2009)	European Union (2013)
	Guatemaia (1985 ⁵ , 2009)	Iceland (2014)
	Argopting (1985°)	Liechtenstein (2014)
	Algentina (1900°, 2001°, 2005°) Rolivia (Plurinational State of) (1999)	NUIWay (2014) Switzorland (2014)
	Ecuador (1988)	Republic of Korea (2016)
	Ecuaudi (1900) Porti (1988, 2012)	Republic of Rolea (2010)
	Venezuela (Rolivarian Ropublic of) (1000)	
	Chile (1993 ^b 2009 2012)	
	Caribbean Community (CARICOM) (1995)	
	Mexico (1995, 2012)	
	Panama (1995)	
	Brazil (1999 ^b 2005 ^b)	
	Cuba (2001 ^b)	
	Paraguay (2005 ^b)	
	Uruguay (2005^{b})	
Costa Rica	El Salvador (1959)	Canada (2002)
	Guatemala (1959)	United States (2004)
	Honduras (1959)	China (2011)
	Nicaragua (1959)	European Union (2013)
	Panama (1962 ^b , 1992 ^b , 2003 ^b)	Singapore (2013)
	Mexico (1984 ^b , 1995)	
	Colombia (1985 ^b , 2016)	
	Dominican Republic (2001)	
	Chile (2002)	
	Caribbean Community (CARICOM) (2005)	
	Peru (2013)	

Table A.3 Selected Latin American countries: intra and extra-regional preferential trade agreements (In chronological order)^a

69

Table A.3 (concluded)

Country	Latin America and the Caribbean	Rest of the World
Mexico	Costa Rica (1984 ^b , 1995, 2012) Panama (1986 ^b) Uruguay (1986 ^b , 2003 ^b , 2004) Argentina (1987 ^b , 2003 ^b) Peru (1987 ^b , 2012) Bolivia (Plurinational State of) (1995 ^b) Colombia (1995, 2012) Venezuela (Bolivarian Republic of) (1995 ^b) Nicaragua (1998, 2012) Chile (1999, 2012) Cuba (2001 ^b) El Salvador (2001, 2012) Guatemala (2001, 2012) Honduras (2001, 2012) Brazil (2003 ^b) Paraguay (2003 ^b)	Canada (1994) United States (1994) European Union (2000) Israel (2000) Iceland (2001) Liechtenstein (2001) Norway (2001) Switzerland (2001) Japan (2005)
Peru	Mexico (1987 ^b , 2012) Argentina (1988 ^b , 2001 ^b , 2006 ^b) Bolivia (Plurinational State of) (1988) Colombia (1988, 2012) Ecuador (1988) Venezuela (Bolivarian Republic of) (1988) Chile (1998 ^b , 2009, 2012) Brazil (1999 ^b , 2006 ^b) Cuba (2006 ^b) Paraguay (2006 ^b) Uruguay (2006 ^b) Uruguay (2006 ^b) Panama (2012) Costa Rica (2013)	Canada (2009) United States (2009) Singapore (2009) China (2010) Republic of Korea (2011) Iceland (2011) Liechtenstein (2011) Norway (2011) Switzerland (2011) Japan (2012) European Union (2013)

Source: Author's elaboration on the basis of information from the World Trade Organization's Regional Trade Agreements Information System (RTA-IS) and the Design of Trade Agreements (DESTA) Database (Dür et al., 2014). ^a Numbers within brackets refer to the year in which the agreement entered into force.

^b Partial scope agreement.





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI).

^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 10.B). Percentages in brackets indicate the share of each origin country/region in 1995.




Figure A.4

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 12.B). Percentages in brackets indicate the share of each origin country/region in 1995.





Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 14.B). Percentages in brackets indicate the share of each origin country/region in 1995.

73





Figure A.6 Colombia: geographical origin of main imported non-primary inputs embodied in gross manufacturing exports, 1995 and 2011^a

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII). Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 16.B). Percentages in brackets indicate the share of each origin country/region in 1995.



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 18.B). Percentages in brackets indicate the share of each origin country/region in 1995.

Trade integration and production sharing: a characterization...

Figure A.8

Mexico: geographical origin of main imported non-primary inputs embodied in gross manufacturing exports, 1995 and 2011^a (*Percentages of the total*)



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 20.B). Percentages in brackets indicate the share of each origin country/region in 1995.



Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Origin sectors are decreasingly ordered according to their share in total imported intermediate goods in 2011 (see figure 22.B). Percentages in brackets indicate the share of each origin country/region in 1995.

Figure A.9

Peru: geographical origin of main imported non-primary inputs embodied in gross manufacturing exports, 1995 and 2011ª

Table A.4 Latin America and the Caribbean: share in imported intermediate goods embodied in Latin American countries' gross manufacturing exports by source industry, 2011ª

(Percentages	of the	total)

Source industry	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Agriculture	Latin America and the Caribbean: 37.2 Brazil (28.9) Paraguay (2.9) Chile (2.0)	Latin America and the Caribbean: 54.6 Argentina (42.5) Paraguay (7.8) Uruguay (2.8)	Latin America and the Caribbean: 64.1 Argentina (44.5) Paraguay (9.7) Brazil (4.1) Guatemala (3.4) Peru (1.6)	Latin America and the Caribbean: 59.8 Argentina (39.2) Brazil (6.2) Ecuador (4.5) Peru (4.0) Paraguay (2.9) Guatemala (1.8)	Latin America and the Caribbean: 15.9 Guatemala (6.6) Nicaragua (2.7) Honduras (1.9) Mexico (1.8) Peru (1.2)	Latin America and the Caribbean: 4.7 Guatemala (1.2)	Latin America and the Caribbean: 46.6 Argentina (34.0) Paraguay (3.8) Brazil (3.6) Guatemala (2.3) Bolivia(Plurinational State of) (1.2)
Mining and quarrying	Latin America and the Caribbean: 88.5 Brazil (85.3) Bolivia (Plurinational State of) (1.0)	Latin America and the Caribbean: 60.7 Chile (45.0) Peru (10.4) Bolivia (Plurinational State of) (2.2) Argentina (1.8)	Latin America and the Caribbean: 82.1 Peru (63.7) Mexico (10.1) Argentina (7.9)	Latin America and the Caribbean: 30.0 Brazil (9.9) Mexico (6.2) Peru (5.8) Chile (1.7) Trinidad and Tobago (1.5) Guyana (1.3) Bolivia (Plurinational State of) (1.2)	Latin America and the Caribbean: 57.0 Guatemala (17.4) Mexico (16.4) Colombia (9.7) Nicaragua (3.3) Chile (2.7) Brazil (2.6) Guyana (1.9) Bolivia (Plurinational State of) (1.2) Peru (1.0)	Latin America and the Caribbean: 19.9 Brazil (7.6) Peru (4.0) Honduras (2.3) Chile (1.7) Bolivia (Plurinational State of) (1.2)	Latin America and the Caribbean: 82.9 Bolivia (Plurinational State of) (34.4) Argentina (22.3) Mexico (16.2) Chile (7.2) Ecuador (1.6)
Food products, beverages and tobacco	Latin America and the Caribbean: 61.6 Brazil (46.7) Paraguay (3.3) Chile (3.3) Uruguay (3.3) Ecuador (2.1)	Latin America and the Caribbean: 36.3 Argentina (19.6) Uruguay (10.3) Paraguay (3.2) Colombia (1.4) Chile (1.0)	Latin America and the Caribbean: 66.9 Argentina (22.1) Paraguay (11.1) Peru (9.6) Guatemala (6.8) Brazil (6.1) Ecuador (3.9) Bolivia (Plurinational State of) (3.5) Mexico (2.0) Colombia (1.3)	Latin America and the Caribbean: 66.8 Argentina (30.5) Bolivia (Plurinational State of) (14.4) Ecuador (9.0) Brazil (5.0) Chile (2.3) Peru (2.0) Mexico (1.3) Honduras (1.1)	Latin America and the Caribbean: 43.1 Nicaragua (15.4) Mexico (8.8) Panama (3.5) Guatemala (3.0) Brazil (2.4) Chile (1.6) Colombia (1.6) Honduras (1.5) El Salvador (1.1) Dominican Republic (1.1) Peru (1.0)	Latin America and the Caribbean: 12.2 Guatemala (3.6) Costa Rica (3.0) Honduras (1.9) Argentina (1.3)	Latin America and the Caribbean: 75.0 Argentina (35.2) Paraguay (12.1) Bolivia (Plurinational State of) (11.2) Chile (4.5) Brazil (3.8) Ecuador (3.4) Guatemala (2.4) Colombia (1.6)
Textiles, leather and footwear	Latin America and the Caribbean: 38.6 Brazil (30.3) Chile (2.3) Paraguay (1.9) Uruguay (1.7) Peru (1.1)	Latin America and the Caribbean: 7.4 Argentina (3.1) Paraguay (1.1)	Latin America and the Caribbean: 18.8 Brazil (5.9) Peru (5.0) Argentina (4.0) Mexico (1.4)	Latin America and the Caribbean: 23.6 Mexico (7.8) Ecuador (4.4) Brazil (4.3) Peru (4.1)	Latin America and the Caribbean: 40.2 Brazil (7.8) Panama (6.9) Argentina (6.2) Mexico (4.6) Honduras (4.1) El Salvador (3.6) Guatemala (3.5) Colombia (2.6)	Latin America and the Caribbean: 8.1 Brazil (2.7) Argentina (1.3) Guatemala (1.1)	Latin America and the Caribbean: 21.8 Colombia (6.9) Brazil (5.1) Argentina (4.1) Chile (2.1) Mexico (1.7)

Source industry	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Wood and wood products	Latin America and the Caribbean: 44.8 Brazil (22.4) Chile (9.5) Paraguay (5.0) Bolivia (Plurinational State of) (3.8) Uruguay (3.7)	Latin America and the Caribbean: 48.1 Argentina (35.0) Ecuador (4.8) Paraguay (3.9) Bolivia (Plurinational State of) (1.8) Uruguay (1.4)	Latin America and the Caribbean: 17.4 Argentina (6.0) Brazil (5.1) Uruguay (2.2) Paraguay (1.6) Bolivia (Plurinational State of) (1.1)	Latin America and the Caribbean: 61.1 Chile (32.5) Ecuador (18.3) Brazil (2.4) Venezuela (Bolivarian Republic of) (2.1) Peru (1.8) Argentina (1.8) Uruguay (1.2)	Latin America and the Caribbean: 40.9 Chile (25.3) Panama (4.8) Guatemala (2.9) Honduras (1.8) Colombia (1.7) Argentina (1.5)	Latin America and the Caribbean: 28.8 Chile (21.2) Peru (3.6) Brazil (1.4) Uruguay (1.2)	Latin America and the Caribbean: 70.2 Chile (44.3) Ecuador (16.6) Argentina (4.8) Brazil (2.1) Uruguay (1.3)
Pulp, paper and paper products	Latin America and the Caribbean: 60.3 Brazil (41.6) Chile (13.8) Uruguay (3.2) Paraguay (1.1)	Latin America and the Caribbean: 13.3 Argentina (8.9) Chile (2.8)	Latin America and the Caribbean: 31.6 Argentina (12.1) Brazil (11.5) Mexico (3.5) Peru (2.3) Colombia (1.6)	Latin America and the Caribbean: 32.7 Chile (15.5) Brazil (8.7) Peru (2.9) Ecuador (2.5) Mexico (2.5)	Latin America and the Caribbean: 31.5 El Salvador (7.0) Honduras (6.5) Mexico (4.7) Guatemala (3.2) Panama (2.6) Chile (2.6) Colombia (2.5) Brazil (1.6)	Latin America and the Caribbean: 3.3 Chile (1.5) Brazil (1.0)	Latin America and the Caribbean: 45.3 Chile (18.1) Brazil (11.4) Colombia (8.4) Argentina (4.9) Mexico (1.7)
Chemical products	Latin America and the Caribbean: 25.6 Brazil (19.7) Mexico (2.8) Chile (1.2)	Latin America and the Caribbean: 10.7 Argentina (4.1) Mexico (2.4) Chile (1.4) Colombia (1.2)	Latin America and the Caribbean: 28.3 Argentina (7.9) Brazil (6.6) Mexico (4.6) Peru (2.7) Trinidad and Tobago (2.4) Venezuela (Bolivarian Republic of) (2.2) Colombia (1.3)	Latin America and the Caribbean: 21.9 Mexico (8.5) Brazil (5.7) Venezuela (Bolivarian Republic of) (2.9) Trinidad and Tobago (1.5)	Latin America and the Caribbean: 24.6 Mexico (9.2) Colombia (4.8) Brazil (2.7) Guatemala (1.9) Panama (1.5) El Salvador (1.3)	Latin America and the Caribbean: 3.1	Latin America and the Caribbean: 24.8 Argentina (6.6) Brazil (5.1) Colombia (4.7) Chile (3.9) Mexico (3.2)
Rubber and plastics products	Latin America and the Caribbean: 51.2 Brazil (40.1) Chile (4.8) Uruguay (2.3) Mexico (1.5) Colombia (1.1)	Latin America and the Caribbean: 22.0 Argentina (8.6) Uruguay (5.6) Colombia (2.2) Mexico (1.7) Chile (1.5) Paraguay (1.5)	Latin America and the Caribbean: 21.3 Brazil (9.7) Argentina (5.1) Colombia (2.0) Peru (1.7) Mexico (1.6)	Latin America and the Caribbean: 30.7 Brazil (8.4) Peru (7.4) Mexico (5.7) Chile (3.1) Ecuador (3.0) Argentina (1.2)	Latin America and the Caribbean: 39.5 Mexico (7.2) Guatemala (6.6) Panama (4.7) Colombia (3.9) Honduras (3.0) Nicaragua (2.9) El Salvador (2.9) Chile (2.8) Brazil (2.3) Peru (1.8)	Latin America and the Caribbean: 3.3 Brazil (1.5)	Latin America and the Caribbean: 26.2 Brazil (7.1) Chile (6.3) Colombia (6.0) Ecuador (2.3) Argentina (1.9) Mexico (1.9)

79

Source industry	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Other non-metallic mineral products	Latin America and the Caribbean: 43.4 Brazil (32.0) Chile (5.3) Mexico (3.1) Paraguay (1.2) Uruguay (1.0)	Latin America and the Caribbean: 13.9 Mexico (6.3) Colombia (2.9) Argentina (2.2)	Latin America and the Caribbean: 37.5 Argentina (15.0) Brazil (8.2) Mexico (6.6) Peru (5.0) Colombia (1.5)	Latin America and the Caribbean: 40.1 Mexico (15.0) Brazil (11.8) Peru (6.6) Ecuador (2.0) Venezuela (Bolivarian Republic of) (1.9) Chile (1.0)	Latin America and the Caribbean: 40.2 Mexico (18.3) Guatemala (6.6) Colombia (4.2) Brazil (3.4) Panama (1.8) Jamaica (1.1)	Latin America and the Caribbean: 5.1 Brazil (3.4)	Latin America and the Caribbean: 29.2 Brazil (8.2) Mexico (7.2) Chile (4.6) Colombia (4.5) Ecuador (1.5) Argentina (1.4)
Basic metals	Latin America and the Caribbean: 62.9 Brazil (52.8) Chile (5.9) Mexico (2.7)	Latin America and the Caribbean: 38.4 Chile (18.7) Peru (5.9) Argentina (4.8) Mexico (3.5) Venezuela (Bolivarian Republic of) (3.4)	Latin America and the Caribbean: 25.3 Brazil (13.4) Argentina (4.8) Mexico (2.5) Peru (2.3) Venezuela (Bolivarian Republic of) (1.6)	Latin America and the Caribbean: 49.6 Mexico (18.5) Brazil (10.9) Peru (7.9) Venezuela (Bolivarian Republic of) (6.2) Chile (3.3) Ecuador (1.1)	Latin America and the Caribbean: 47.6 Chile (11.7) Mexico (10.0) Brazil (6.2) Guatemala (4.3) Trinidad and Tobago (3.3) El Salvador (3.2) Venezuela (Bolivarian Republic of) (3.0) Dominican Republic (2.5)	Latin America and the Caribbean: 7.1 Chile (2.7) Brazil (2.2)	Latin America and the Caribbean: 28.4 Brazil (12.6) Mexico (4.9) Chile (4.2) Argentina (1.8) Trinidad and Tobago (1.8) Venezuela (Bolivarian Republic of) (1.6)
Fabricated metal products	Latin America and the Caribbean: 32.1 Brazil (24.9) Chile (5.5)	Latin America and the Caribbean: 7.9 Argentina (3.5) Chile (2.6) Mexico (1.2)	Latin America and the Caribbean: 14.3 Brazil (4.7) Argentina (3.4) Mexico (2.6) Peru (2.3)	Latin America and the Caribbean: 14.7 Mexico (4.5) Brazil (4.4) Chile (1.7) Venezuela (Bolivarian Republic of) (1.2) Peru (1.1)	Latin America and the Caribbean: 28.1 Mexico (8.7) Guatemala (4.1) Brazil (3.5) Colombia (3.0) Panama (2.2) Honduras (1.8) El Salvador (1.4) Dominican Republic (1.1)	Latin America and the Caribbean: 1.2	Latin America and the Caribbean: 16.5 Chile (4.9) Brazil (4.6) Mexico (2.2) Colombia (2.0)
Machinery and equipment, n.e.c.	Latin America and the Caribbean: 15.9 Brazil (13.5) Chile (1.2)	Latin America and the Caribbean: 3.6 Argentina (2.3)	Latin America and the Caribbean: 7.1 Brazil (4.3) Argentina (1.3)	Latin America and the Caribbean: 10.6 Brazil (4.1) Mexico (2.1) Argentina (2.0) Chile (1.2)	Latin America and the Caribbean: 12.5 Mexico (4.8) Brazil (3.1) Colombia (1.7)	Latin America and the Caribbean: 1.8 Brazil (1.5)	Latin America and the Caribbean: 12.2 Brazil (4.5) Chile (3.1) Mexico (2.1) Argentina (1.2)
Computer, electronic and optical equipment	Latin America and the Caribbean: 15.2 Brazil (7.4) Mexico (6.4) Costa Rica (1.1)	Latin America and the Caribbean: 3.8 Costa Rica (2.2) Mexico (1.1)	Latin America and the Caribbean: 12.2 Mexico (5.9) Brazil (4.9)	Latin America and the Caribbean: 14.6 Mexico (7.5) Brazil (2.8) Ecuador (2.2) Panama (1.1)	Latin America and the Caribbean: 1.3	Latin America and the Caribbean: 3.5 Costa Rica (3.4)	Latin America and the Caribbean: 15.3 Mexico (5.2) Brazil (3.2) Costa Rica (2.1) Chile (1.9) Panama (1.5)

Table A.4 (concluded)

Source industry	Argentina	Brazil	Chile	Colombia	Costa Rica	Mexico	Peru
Electrical	Latin America and the						
machinery and	Caribbean: 34.1	Caribbean: 4.3	Caribbean: 17.3	Caribbean: 24.5	Caribbean: 24.1	Caribbean: 1.3	Caribbean: 20.8
apparatus, n.e.c.	Brazil (30.0)	Argentina (2.0)	Brazil (5.5)	Mexico (11.8)	Mexico (12.7)		Colombia (5.8)
	Uruguay (2.0)	Mexico (1.8)	Mexico (4.9)	Brazil (6.2)	Panama (2.5)		Brazil (4.1)
	Mexico (1.4)		Argentina (3.5)	Chile (3.4)	Brazil (2.4)		Chile (4.1)
			Colombia (2.0)		Colombia (1.8)		Mexico (4.1)
					Peru (1.6)		Argentina (2.1)
Motor vehicles	Latin America and the						
	Caribbean: 62.8	Caribbean: 16.2	Caribbean: 26.8	Caribbean: 18.4	Caribbean: 17.4	Caribbean: 3.5	Caribbean: 30.6
	Brazil (59.6)	Argentina (12.4)	Brazil (17.7)	Brazil (12.6)	Brazil (11.5)	Brazil (3.1)	Brazil (23.6)
	Mexico (1.5)	Mexico (2.6)	Argentina (5.3)	Chile (2.2)	Mexico (2.5)		Mexico (4.1)
	Uruguay (1.1)	Chile (1.0)	Mexico (1.9)	Mexico (2.2)	Panama (1.2)		Chile (1.5)
			Panama (1.0)				
Other transport	Latin America and the						
equipment	Caribbean: 7.3	Caribbean: 2.2	Caribbean: 9.9	Caribbean: 1.5	Caribbean: 5.0	Caribbean: 1.9	Caribbean: 1.5
	Brazil (6.9)	Mexico (1.7)	Brazil (7.5)	Brazil (1.2)	Colombia (2.0)	Brazil (1.9)	
			Argentina (1.9)		Panama (1.3)		
					Brazil (1.0)		

Source: Author's calculations on the basis of current-price input-output data from the 2016 edition of the Organisation for Economic Co-operation and Development (OECD)'s ICIO tables; and trade data from the Centre d'Études Prospectives et d'Informations Internationales (CEPII), Base pour l'Analyse du Commerce International (BACI). ^a Only countries with a share larger or equal than 1% are listed in the table.



Series:

International Trade

Issues published

A complete list as well as pdf files are available at

www.eclac.org/publications

- 137. Trade integration and production sharing: a characterization of Latin American and Caribbean countries' participation in regional and global value chains, Dayna Zaclicever, (LC/TS.2017/161), 2017.
- 136. Use of knowledge intensive services in the Chilean wine industry, Fulvia Farinelli, Karina Fernández-Stark, Javier Meneses, Soledad Meneses, Nanno Mulder and Karim Reuse, (LC/TS.2017/147), 2017.
- 135. Opciones para la convergencia entre la Alianza del Pacífico y el Mercado Común del Sur (MERCOSUR): la regulación de la inversión extranjera directa, Sebastián Herreros y Tania García-Millán (LC/TS.2017/81), 2017.
- Ajustes en producción y empleo ante choques de magnitud al comercio: evidencias al nivel de firmas en Colombia, Lucas Navarro y Andrea Pellandra, (LC/TS.2017/79), 2017.
- El impacto del comercio con China en los mercados laborales locales de Chile, Andrea Pellandra, (LC/TS.2017/54), 2017.
- Asia-Pacific and Latin America: dynamics of regional integration and international cooperation, Akio Hosono, (LC/TS.2017/49), 2017.
- La irrupción de China y su impacto sobre la estructura productiva y comercial en América Latina y el Caribe, José Durán Lima y Andrea Pellandra, (LC/TS.2017/6), 2017.
- 130. El acuerdo de Asociación Transpacífico (TPP) y el agro, Daniela Alfaro, (LC/TS.2017/4), 2017.
- 129. Crisis y debates sobre globalización en Europa y Estados Unidos: implicancias para América Latina, Osvaldo Rosales, (LC/L.4285), 2017.

137

TRADE

INTERNATIONAL TRADE



ECONOMIC COMMISSION FOR LATIN AMERICA AND THE CARIBBEAN COMISIÓN ECONÓMICA PARA AMÉRICA LATINA Y EL CARIBE www.eclac.org